

Therapy Web User Manual

Version 1.0

7/8/2025

Ascom UMS s.r.l. Unipersonale Via Amilcare Ponchielli 29, 50018, Scandicci (FI), Italy Tel. (+39) 055 0512161 – Fax (+39) 055 829030 www.ascom.com

Contents

Therapy Web	4
1. Introduction	4
2. Module selection	4
2.1. Patient selection	4
3. Basic concepts	5
3.1. Prescription status	5
3.2. Repeatable vs. Not Repeatable Prescriptions	
3.3. Punctual vs. Durative administrations	7
3.3.1. The therapy cycle - treatment plan re-confirmation	7
3.3.2. Orders generation	
3.3.3. Orders validity	8
3.3.4. Order validity expiration	8
4. The "Therapy Prescription" module	9
4.1. Module selection	9
4.2. Prescription main page	9
4.3. History	11
4.4. Edit a prescribed treatment	12
4.5. Suspend a prescribed treatment	14
4.6. Remove a prescribed treatment	15
4.7. Skip or Execute an expired prescribed order	15
4.8. Prescribe changes to a running administration	
4.9. Indicators on prescriptions	
4.10. Therapy Cycle	
4.11. Filters	19
4.11.1. Filters - Categories	20
4.11.2. Filters - Routes	
4.11.3. Filters - Forms	
4.11.4. Filters - Status	
4.12. Grouping	
4.13. Command Bar – Action buttons	
4.14. External resources (Help)	
4.15. Create a New Prescription	24
4.15.1. New Prescription – Active Moieties Quantity/Dose step	
4.15.2. New Prescription – Range specification for Quantity/Dose	
4.15.3. New Prescription – Scheduling step	
4.15.4. New Prescription – Review/Confirm step	53

	4.16. Prescription with double signature	54
	4.16.1. Signature on a different Digistat session	55
	4.16.2. Signature on the same Digistat session	57
	4.17. Overview the created Prescriptions	61
	4.17.1. Overview – Timeplan view	62
	4.17.2. Overview – Summary view	64
	4.18. Update the Therapy Cycle	67
	4.19. Closed Therapy	69
	4.20. QR code display	69
5. 1	The "Therapy Execution" module	71
	5.1. "Therapy Execution" module selection	71
	5.2. Main screen	71
	5.3. Prescribed orders representation	71
	5.3.1. Colors legenda	72
	5.4. The orders chart	73
	5.4.1. Graphic representation of the orders	73
	5.5. How to record the treatment administration	75
	5.5.1. Punctual treatment administration	75
	5.5.2. Durative treatment administration	77
	5.5.3. Drugs interactions	80
	5.5.4. Allergies	80
	5.5.5. End a durative administration before or after the scheduled time	e81
	5.5.6. Changing the administration values	81
	5.5.7. Prescribed changes of a running administration	83
	5.5.8. Administration of an order either expired or not-yet-validated	
	5.5.9. Skip administration	
	5.5.10. External resources (Help)	
	5.6. "Range" administration	
	5.7. Administration with double signature	
	5.7.1. Signature on a different Digistat session	
	5.7.2. Signature on the same Digistat session	
	5.8. The administration window	
	5.9. "Execution" module prompts	
	5.10. "Execution" time range display	
	5.11. Treatment sorting buttons	
	5.12. Compact view	
	5.13. Closed Therapy	. 107

Therapy Web



For information about the Product environment, precautions, warnings and intended use see USR ENG Digistat Care and/or USR ENG Digistat Docs (depending on the modules installed - for the Digistat Suite EU) or USR ENG Digistat Suite NA (for Digistat Suite NA). The knowledge and understanding of the appropriate document are mandatory for a correct and safe use of Therapy Web, described in this document.



The treatments displayed in the figures of this manual are examples created to better explain the procedures of Therapy Web. Their features (as, for example, dosages and names) are not intended to be in any way clinically relevant.

1. Introduction

Digistat[®] "Therapy Web is a web application that can be of help for the clinical staff for the documentation of the prescription and administration of treatments.

2. Module selection

Digistat[®] "Therapy Web" is formed of two modules: "Therapy Prescription" for the treatment prescription documentation, mainly used by physicians; "Therapy Execution" for the treatment administration documentation, mainly used by the nursing staff.

On the lateral bar:

- > Click the icon to launch "Therapy Prescription".
- Click the icon to launch "Therapy Execution".

2.1. Patient selection

If no patient is selected, no data is displayed.

To select a patient,

click the Patient button on the Control Bar (Fig 1)

The "Patient Explorer Web" module opens. See the "Patient Explorer Web" module user manual for the operating instructions (*USR ENG Patient Explorer Web*).



Other modules can be configured for the patient selection in place of "Patient Explorer Web", depending on the configuration of the Digistat Suite. If this is the case, see the specific documentation for instructions.

When a patient is selected the patient's name is displayed on the **PATIENT** button instead of "**Choose Patient**".

The "Therapy Prescription" and "Therapy Execution" modules display the data of the selected patient.

3. Basic concepts

This section explains some basic notions whose preliminary understanding is essential to the appropriate use of "Digistat Therapy".

3.1. Prescription status

The "Therapy Prescription" module (described in section 4) allows to create a treatment plan and keep it active in time after periodic confirmations. Possible variations can be recorded quickly while the general plan remains the same.

The patient's treatment plan is summarized and displayed in a table (Fig 2). Each row corresponds to a treatment prescription.



Fig 2

Each prescription generates a certain number of orders. The orders correspond to the single administrations of the treatment.

A prescription is terminated when all the orders that it generated are executed and no other order will be generated by it in the future.

The prescription status is displayed on the right (Fig 3).



There are four possible prescription statuses:

- 1. "Active" when a prescription is in "Active" status, the prescription values can be displayed and edited. Click the corresponding row to open the related "prescription specification" window.
- 2. "Terminated" a prescription is terminated when all the orders that it generated are executed and no other order will be generated by it in the future. Terminated prescriptions are still active. That means that the prescription values can be displayed and edited. Double-click the row corresponding to the prescription to expand the row and display the related "prescription specification" tools.
- 3. "Suspended" the prescription suspension, performed using the **Suspend** button (see section 4.5), deletes the prescription from the prescription plan and deletes all the related administration orders. The suspended prescriptions can be displayed again (in strikethrough text) by means of the available filter (see section 4.11 for the filters explanation). Suspended prescriptions can then be resumed (section 4.5). A suspended prescription does not generate administration orders.
- 4. "Removed" a prescription can be removed by means of the **Remove** button (see section 4.6). All the administration orders generated by it are deleted. The removed prescriptions can be displayed again (in strikethrough text) by means of the available filter (see section 4.11 for the filters explanation). Removed prescriptions do not generate administration orders and cannot be resumed.

3.2. Repeatable vs. Not Repeatable Prescriptions

Repeatable prescriptions

A prescription is "*Repeatable*" if it generates orders that must be administered at given intervals. For example: a prescription can state that a drug must be administered every day at a certain time. In this case the prescription is repeatable; it generates the corresponding orders placing each of them in the required days at the required time/s.

Non-repeatable prescriptions

Non-repeatable prescriptions generate only the administration orders explicitly specified in the prescription. When all the orders are executed, the prescription is "Terminated" (see section 3.1).

The repeatability of a prescription is set by a specific checkbox on the prescription specification window (see Fig 76 for an example).

The repeatability of a prescription is indicated by a specific icon - $\stackrel{\square}{\rightarrow}$ - on the prescriptions table of the "Therapy Prescription" module (see, for example, Fig 4 A).



Paracetamol — Soluble tablet DOSE: 1000 mg — Oral — 2 times a day at 11:00 AM · 5:00 PM — until further notice

Fig 4

3.3. Punctual vs. Durative administrations

Punctual administrations start and end in one moment (it's the case of a tablet, for instance).

Durative administrations last a certain amount of time (it's the case of a drip, for instance).

Punctual and durative administrations are characterized by different graphic features and different management procedures on the Therapy Execution module. See section 5.3 for the description of the possible administration types on Therapy Execution.

3.3.1. The therapy cycle - treatment plan re-confirmation

The validity of the treatment plan created on "Digistat Therapy" is limited in time and needs to be re-confirmed by the medical staff. The treatment plan validity period is called "Therapy Cycle". The "Therapy Cycle" re-confirmation is a safety procedure that forces to check and verify the treatment plan at specified intervals. The "Therapy Cycle" duration is configurable. See section 4.18 for the therapy cycle update procedure.

3.3.2. Orders generation

This paragraph explains how the system generates the orders from a prescription specified in the treatment plan.

At prescription time, i.e. when the **Save** button on the "Summary" screen is clicked (Fig 94 **A**, see paragraph 4.15.4), the system generates the orders corresponding to the prescription specifications.



Conditional prescriptions (paragraph 4.15.3.2) do not generate orders. These prescriptions are executed only when certain specific conditions occur.

3.3.3. Orders validity

The order execution standard procedure requires the order validation before the execution. The orders that, when generated, are within the "Therapy Cycle" are automatically validated.

The other orders are validated every time the "Therapy Cycle" is updated when they are within this period (paragraph 4.18).

Non-validated orders are signaled by a specific icon - - on the Therapy Execution module (Fig 6). They can be executed only after a specific procedure, described in paragraph 5.5.7.



See paragraph 3.3.1 for the explanation of the "Therapy Cycle" (or "treatment plan validity period").

See paragraph 4.18 for the treatment plan update procedure.

3.3.4. Order validity expiration

The validity of an order expires after a certain amount of time after the scheduled administration time has passed. That is: if a validated order is not administered at the scheduled time, it remains validated for a certain period. After this period the order goes back to non-validated state (it is named "expired" to differentiate it from future not-yet-validated orders). Expired orders can be deleted from the treatment plan through the "Therapy Cycle" update procedure. See paragraph 4.18. Expired orders can be executed only using a specific procedure. The procedure is described in paragraph 5.5.7.



The orders validity duration is defined by a specific configuration parameter. Refer to the system administrator for more information.



The validity of a conditional prescription expires when the "Therapy Cycle" expires. Thus, conditional prescriptions are automatically validated every time the "Therapy Cycle" is updated. See paragraph 4.15.3.2 for the explanation of "Conditional prescriptions".

4. The "Therapy Prescription" module

4.1. Module selection

To select the "Therapy Prescription" module:

click on the <a>icon on the lateral bar provided on the left:

When a module is selected the corresponding icon is highlighted. When a patient is selected, it is possible to access to his or her prescription page and the "Therapy Prescription" module's main screen opens.

4.2. Prescription main page



Fig 7

When a patient does not have any prescribed therapy, the page is empty and the message "**No Therapy Prescribed**" (Fig 7 **A**) is displayed in the central part of the page.

If some therapies have been prescribed, the patient's therapeutic plan is shown and the treatments listed in the central part of the page (Fig 8 **A**), together with information on their modes of administration, dosages, schedule, and status.

2 174.1111	House or druct	
🖌 Dus al times	Taken S. Laker S.	T #1
Andgewis Newsieland and a Rosenwary	Firsting - minutened and Frank	
	B-WARY	
	Dooms Stateman III. Milling	
	and been and here and and and and and and the test that the test that the test and	per ti gris
	Transversional product test Aug	where the extension of the
	Next Sec.	
	Incl. 30 Miles (2017 - Section 1970) - Representant (2016 - Section 1970) - Section 1970	e =
	RANGETRANCED - 0010440110 10105 - 5000 erg = 105017 - comis a situa at 1000 - cello face (0.0000 - cello factore review)	and the second
T Seed		
an angener		
er dans		
an Anna		
18 August	0	
0.000		

Each row in the list corresponds to a treatment prescription and the prescriptions are managed and displayed according to the user preferences.

The **name** of the administered drug/product name, together with information on its **mode of administration**, **dosages** (first element displayed on the second row of the prescription string) with **solution speed/amount/volume information**, **routes of administration/infusion** (if present) and **scheduling** (date and time) of prescribed order/s are displayed on each prescription row.

		v	V
Ibuprofen – Medicated plaster DOSE: 0.2002 g – Oral – 11/29/2023, 11:08 AM		TEMMANATED	*
Peroxetine — -Dral-suspension DOSE: 200 mg — Oral — a condition		TUNPENDED	*
Paracetamol — Soluble tablet DOSE: 200 mg — Oral — 2 times a day at 3:00 PM - 5:00 PM — until 11/29/2023, 5:00 PM	C 1	ACTIVE	~
Ketoprofen — Capsule BOSC: 200 mg — Gral — Thursday, Friday at 400 PM - 6:00 PM — effective; 11/30/2023, 4:00 PM — uniti 12/1/2023, 6:00 PM	tı	REMOVED	×
Dopamine — Solution for injection DOSE: 20 ml at 2 ml/h — Intravenous — 5 times in 10 min — until 11/29/2023, 11:08 AM	C	ACTIVE	*

Fig 9

The **status** of each prescription is displayed on the right part of the prescription row (Fig 9 **A**).

The following statuses can be retrieved and shown:

- Active (enabled by default);
- Suspended;
- Removed;
- Terminated.

		TUNENARY										DANNAR .					
	20 ml		an a	skart 15 kg						-254	оне 4.43 ml	1.1	антан 1h, & 36	n.			
€ 17444 8 AMA 17 AMA 18 AMA 19 A	w 1944	11.844 12.944	D1 PM	12.944	05.750 5	ANY 15	37.5%	03 PM	₩ PM	10.966	11.14	72.844	31 AM	02.008	111.000	(+ AM	25.444
Mur 11 Delember												C	>	solut h		Dia ini	(A), 223
HISTORY													e	nn -	305/6	ND .	REMOV

A down-arrow button (Fig 9 B) is available to expand a **detailed section** where a summary (Fig 10 **A**) of the treatment and its prescription values (amount, volume, speed, duration, patient weight), as well as information on the ongoing therapy and prescribed scheduling are shown (Fig 10 B). Information on the **user** who created and saved the prescription together with the **date-time** the prescription was last modified is also provided (Fig 10 C). Finally, several buttons are available depending on the current status of the prescriptions (Fig 10 D).

By default, on an active prescription, the following buttons are provided:

- **History**: to access the prescription history information;
- Edit: to edit the prescription;
- **Suspend**: to temporarily stop the prescription;
- **Remove**: to permanently remove the prescription (it cannot be resumed).



Clicking the **History** button, a modal window opens (Fig 11) and the prescription module in the background is dimmed.

The **name of the treatment/drug** is placed on the window header (Fig 11 **A**), then the prescription string (or prescription strings if the prescription has been modified) is shown (Fig 11 **B**).

Date and time of the prescription creation is retrieved together with the username of the user who created the prescription (Fig 11 C).

The button **Close** (Fig 11 **D**) is present in the right-bottom part of the window to close the History view.

Sections for each treatment day of the selected prescription are displayed and the actions performed on the prescription are recorded, listed, and grouped by the date on which they were performed (Fig 11 **E**).

9	>	Tue, Dec 12, 2023, 2:16 PM CREATED by ADMIN	
12/12/2023	330 PLANNED 152	121222223, 2-07 PM OREOWED BY ADMIN	10 mg
	1100 PLANNED PS2	12/13/2023 10 Sc AM EXECUTED BY ADMAN	10 mg
12/13/2823	Too PLANNED AM	12/12/2021 11:51 and CAWCELLED BY ADMIN	
	ton manufacture Pad		D 10 mg
		Fig 12	

Each section contains:

- Date and time of originally planned administrations (Fig 12 A);
- Date and time of the performed actions on the prescription orders (Fig 12 B);
- The action performed and the name of the user who performed it (Fig 12 C);
- The dosage of the administrated prescription order (Fig 12 D).

4.4. Edit a prescribed treatment

	IEPARAZIONE INIETTABILE ILE DILUITO — 4/15/2024, 2	54 PM			Communication Communication
			SUMMARY		
			миолят wesнт 000 mg 75 Kg		
12 AM Wed 17 April	04 AM	OR AM	12.PM	D4 PM	OB PM
reatment started M	on, Apr 15, 01:21 PM		2	Prescr	ibed by ADMIN Mon. Apr 15, 02:54
				A EDIT	SUSPEND REMOVE

Clicking on the **Edit** button (Fig 13 **A**), the **Edit window** of the selected prescription is displayed.

Quantity/Dose			dukr		Review Confirm
			EPARAZIONE INIETTABILE ILLATO — 4/15/2024, 2:54 FM		
PREPARAZIONE IN	E INIETTABILE		ORALE DILUITO		
	Amount 1000	:	Weight 75	:	
	mg		Ка		

Fig 14

The Edit window opens on the **Quantity/Dose** step (Fig 15 **E**) that will be described later on in paragraph 4.15.1.

On the left, a lilac bar with a button is provided to expand a section (Fig 14 **A**), which is compressed by default, containing therapy information.

> Click on the available button to expand the section.

CURRENT THERAPY	Quantity/Dose	Schet		 Review/Continn
MODICILLINA - COMPRESSE DISPERSIBILI OSE: 500 mg - CRALE - 4/15/2024, 12:17 PM		PARACETAMOLO - PRE	PARAZIONE INIETTABILE LUITO 4/15/2024, 2:54 PM	
URDSEMIDE - 1 mg/ml - SOLUZIONE PER INFL/SIONE IOSE: 1060 mg in 1000 ml ac 698 mL/h (690 mg/h) - ENDOVENIOSA DILUXTO - 4/17/0024, 10.08 /WI	Famil	anan ananan se d	faulte	
ARACETAMOLO – PREPARAZIONE INIETTABLE NOTE: 1000 mg – ORALE DILUITO – 4/15/2028, 2:54 PM	PREPARAZION	E INIETTABILE 🔹	ORALE DILUITO	
	776925	Amaunt	Weight	
	1000	:	75	
	mg		Kg	
	۵ (
НІЗТОМУ	© ۱			
	<u>ہ</u>			
VIRCIDIA 2007/W - TOOD ING - DIVALE DILLATO	<u>ہ</u>			
ANICODA 200791 - 1000 mg - DRALE DILLITO	Ĉ			CANCEL

On the left, the section opens, and it consists of two parts:

- The **Current Therapy** portion (Fig 15 A) containing all the prescriptions created and associated to the selected patient (including the selected to be edited one) and
- the **History** portion (Fig 15 B), containing a history summary of the actions performed on the selected prescription.

Both cannot be modified, and they appear in read only mode.

The originally prescribed values, form and route of administration are retrieved and shown. It is not possible to change the form and the route of administration, but only the prescribed values (Fig 15 C). It is not possible to change the units of measure originally set. To edit the values:

- Replace the values to be changed by overwriting the new value or by using the up and down arrows to increase or decrease the original value by one unit at a time.
- Click the NEXT button (Fig 15 D) to move to the Schedule step (Fig 15 F). Some modifications on the schedule settings can be performed but it is not possible to opt for a totally different schedule and change it. More information on prescription scheduling will be provided in paragraph 4.15.3.
- Click the NEXT button to move to the Review/Confirm final step (Fig 15 G). The summary is provided, and it shows the edited values. A note field is given to enter a note to the Editing procedure. More information on prescription Review step will be provided in paragraph 4.15.4.
- Click the Update button to save the edited prescription and update it accordingly or on **Cancel** button to abort the editing procedure.

It is also possible to click on **Back** button to go back to **Schedule** and **Quantity/Dose** step.

4.5. Suspend a prescribed treatment

To suspend a prescribed treatment:

Click the Suspend button.

A popup appears asking the user to confirm the action.



Once the prescription of the selected treatment is stopped, it disappears from the list displayed on the main page or, in case the suspended prescription are not filtered out (see paragraph 4.11.4), the entire prescription string appears in strikethrough font (Fig 16 A) and the Status is set to "Suspended" (Fig 16 B).

		SUM	volity.			
		amourt 23 mg	mod BD Ag			
	10 mg	ta ng	1			
DE AM Nel 13 December	13 PM	04.954	DE PM	12 AM Thu 14 December	164 AM	
				Areaso ibed	ay adamwyna wind y	Nec (18, 195,27)

Exploding the suspended prescription row again, the **Resume** button (Fig 17 **A**) is present instead of the **Edit** and **Suspend** ones. To resume the administration of the prescribed treatment:

> Click on the **Resume** button provided.

4.6. Remove a prescribed treatment

To remove an administered treatment:

Click the **Remove** button.

A popup appears asking the user to confirm the action.





Once the prescription of the selected treatment is removed, it disappears from the list displayed on the main page or, in case the removed prescription are not filtered out (see paragraph 4.11.4), the entire prescription string appears crossed out and written in light grey characters (Fig 18 **A**), and the **Status** is set to "**Removed**" (Fig 18 **B**).

4.7. Skip or Execute an expired prescribed order



Fig 19

When a prescription contains an expired order, which has not been administered according to the established treatment plan, the prescription is highlighted with a red colored bar (Fig 19 **A**). Exploding the prescription row, two buttons are provided:

• Skip button (Fig 19 B) to skip the administration of the expired treatment or

• **Execute** button (Fig 19 C) to execute the administration immediately.

For the administration to be successful, it is essential to enter a note justifying the delay in execution.

Methodole — Calutole 00/02: 0.075 mg — Draf — write a day at 1100 AM — unit 12/03 (2002, 7:00 AM	0 3	10708	~
Glacter - Musici 2012: 100 mg in 1000 ml at 1 mg/h - Route of administration not applicable - Mundey, Technoley, Selundey, Wednesday at 10 60 MM - Level (2/62003, 10:00 MM	e	scrus	~
Bes venues — Provider and solvers for autocon for repetion DMMP, 20 m6 at 10 m8/h — Socialametric — 2 things in 24 h — until 1211 (2023, 0:05 AM	: e .2 ₀	40716	



4.8. Prescribe changes to a running administration

It is possible to prescribe changes or to generate a "Stop" order to durative administrations while they are running (see sections 3.3 and 5.3 for the description of durative administrations).

To do that:

Click the row of the prescription that generated the order for the running administration.

The prescription row enlarges to display the prescription buttons, summary, administration values and chart (Fig 21).

JROSEMIDE — 1 OSE 1000 mg in 1	mg/ml — SOLI 1000 ml at 100	JZIONE PER INFUSI 0 ml/h (1000 mg/h)	- ENDOVENOSA	DILUITO - 4/17/	2024, 12:13 PM				ACTIVE today	^
		SUMMARY					RUNNING			
MOUNT 1000 mg	vouve 1000 ml	EDISCENTIVITION 1 mg/ml Dunancis web 1h 90		okus iPito 1000 mg/h		-7,974 mg	võruwé ~7.974 ml	OLSIATION ~		
					1000 11					
2 AM Wed 17 April	04 /	AM	OS AM		M5 51	04 PM		08 PM		
Treatment startes	d Wed, Apr 17,	12:13 PM			A		Prescribe	d by ADMIN Wed,	Apr 17, 12:1	13.PM
HISTORY					$\mathbf{\mathbf{v}}$		EDIT	SUSPEND	REMO	VE.
				F	ia 21					

Two additional buttons are on the window (Fig 21 **A**).

"Change" order

 \succ Click the $\hat{}$ button to prescribe a change to the current administration values.

The following window is displayed (Fig 22).

1000	Speed •	Drug speed 1000	:
ml/h		mg/h	
	RUNN	No	
	AMOLAST (CO.	AND DURATION	
	-124 mg -12	I.mî −7m	
	Autor		



Click the Update button (Fig 22 B).

A "Change" order is generated on the Therapy Execution module. This case is described in paragraph 5.5.7.

"Stop" order

> Click the button to prescribe to stop the current administration.

The following window is displayed (Fig 23).



Click the **Stop** button (Fig 23 **A**).

A "Stop" order is generated on the Therapy Execution module. This case is described in paragraph 5.5.7.



The buttons indicated in Fig 21 A, - 👻 and 📕 - generate a "Change" or "Stop" order that refers to the current running administration. The **EDIT** button, described in paragraph 4.4, changes the values of the overall prescription and generates a new set of orders aligned to the updated values.

4.9. Indicators on prescriptions

Finally, different icons/indicators can be displayed next to the status label, which can indicate:

- the current prescription status related to the prescription itself (like the repeated schedule $\stackrel{\leftarrow}{\leftarrow}$), the orders, the active infusion (if any);
- if the prescribed drug has interactions with other drugs prescribed within the same
- treatment plan 💜 :
- if the prescribed drug is allergic to the patient, as indicated in the patient's file
- if the treatment was prescribed urgently and the chosen scheduling was "Emergencv"

- changes in the **patient weight** \square : if the patient weight set in the prescription does not match with the one retrieved from the system the scale icon is displayed;
- differences between:
 - the prescription of a drug and its administration in time (execution clinical time changed Second administration time outside of the prescribed range 🕓) or
 - \circ the dose values administered \blacksquare or the infusion values administered \checkmark and the values originally prescribed.

4.10. Therapy Cycle



Fig 24

On the central part of the page, on top of the prescription list, the **therapy cycle status box** is provided, with updated information on the date of the last therapy review and on the next update due. A green horizontal bar is available indicating the progress of the current therapy. The size of the bar grows proportionally to the elapsed therapy time.

REVIEWED	NEXT UPDATE	
5 days ago	4 days ago	
 Elec (
Fig 2	25	

Once the therapy has expired, the completion tracking bar turns red.

4.11. Filters

The created prescriptions can be filtered using different provided filters and combinations of those available filters. On the left part of the page, the filters section is provided (Fig 8 **A**).

		10,0343
	2	State of classes
<u> </u>	•	Anapasia
	0	Nonserviced and inflationality
₿	W	Stop at
	-	Acamin
	-	Barne
	-	Farm
	3m	Geograph
	•	See.
		Fig 26

The "**Show all classes**" (Fig 26 **A**) and "**Show all**" (Fig 26 **B**) buttons are selected by default, and they can be also used to remove all the filters applied. This way, all prescriptions are displayed in the central part of the prescription view. When corresponding filters are selected, the "Show All" and "All ATC Groups" buttons are automatically deselected.

4.11.1. Filters - Categories

Each treatment can be associated to a single class (the therapies can be grouped under a customizable number of classes) and to one or multiple routes of administration and forms by configuration. So, it is possible to filter the treatments by their Category, Routes and Forms. Moreover, it is possible to filter the treatments by their Status.

C. Frankline	anti-anti- Aguarda againta		T #	0
Andreas		÷ @	and the second	ĩ
	аналанан на саланан на саланан на саланан на салан на с	ę	-	1

A button is provided for each class. To filter the treatments according to their Class:

Click on a colored button corresponding to a class (Fig 27 A) to display on the central screen only those treatments corresponding to that class (Fig 27 B).

(B)

It is possible to select one or more classes: the treatments associated to those classes will be all displayed. Once selected, the buttons are highlighted, and a check mark symbol appears in the radio button provided for each class.

- Click a second time to deselect the selected class.
- > Click on the "Show all classes" button (Fig 27 C) to display all prescribed treatments belonging to all available classes.

Only the classes to which the actual prescribed treatments belong are displayed in the filter section and not all available and configured ones.

4.11.2. Filters - Routes



Fig 28

It is also possible to filter the prescribed treatments according to their Routes of administration:

- Click on Routes filter button (Fig 28 A) to open the menu where available routes (only those configured for the prescribed therapies) are collected and can be selected (Fig 28 B).
- Select one or more routes.

Prescribed treatments are filtered on the page according to the selected routes.

Click on the funnel button provided (Fig 28 C) to remove the applied filter/s.

The list of routes configured and present in all prescriptions associated with the selected patient and only those routes (not all possible routes) is displayed.

4.11.3. Filters - Forms

••• Routes	0.	Infusion	Capsule	Tablet	
··· Forms					220100000000
(≡ Grouping	Powd	er and solve	nt for solutio	n for injection	Effervescent tat



It is possible to filter the prescribed treatments according to their Forms:

- Click on Forms filter button (Fig 29 A) to open the menu where available forms (only those configured and selected for the prescribed therapies) are collected and can be selected (Fig 29 B).
- Select one or more forms. Prescribed treatments are filtered on the page according. to the selected forms.
- Click on the funnel button provided (Fig 29 C) to remove all the applied filters.

The list of forms configured and present in all prescriptions associated with the selected patient and only those forms (not all possible forms) is displayed.

4.11.4. Filters - Status

It is possible to filter the prescribed treatments according to their changing **Status**:

- > Click on Status filter to open the menu where available actual statuses can be selected. The following prescription statuses can be displayed:
 - **Active**, that is selected by default;
 - Suspended (only if there are suspended prescriptions in the treatment list/plan);
 - **Removed** (only if there are removed prescriptions in the treatment list/plan);
- Select one or more statuses.

Prescribed treatments are filtered on the page according to the selected statuses.

Click on the funnel button provided to remove all the applied filters.

4.12. Grouping



It is possible not only to filter but also to group the treatments on the main screen.

Click on the Grouping button (Fig 30 A) and then choose between the three available options:

HC 1985	Artes around			
Star of Council	A provida - a giorra fa			
Andgene:		÷ e	ALCONO.	i.
) Mantanan Kapi Sama aki asi damany			-	٣
	224-58.058 51 Topics 111111-102 regist Topics All Star shifts (15 mg/c) - Tool			۴
	18.9425441 Aus Static.come DOI:158-mg - (340054-# 1000 - 1040) A 2042/25.#(0300 - 0405 - 1000 - 2042) And Color - 5040	Ð	-	i*
	ASTRUCTURAL ANSWER HER SE - SOLUCIONE RETINING. TOTAL	00		.~
2 mail	1006 3 ml at 100 ml/h - ENDORSNOSA DILUTID - 344/204-11:02	er -	13 8441	
- August	and relations and the relations	(D)	-	~
there .	AC100 ACTREMERICO - COMPETER DOGL 180199 - CHALE DOUND - ENGEDE & TOD- ENGEDE & TOD- SHEET ALSO	6 0		٣
tere :	Coversionale	C 2	a.ettantii	Ψ.
t from the				
- heat				



- Class: to group treatments according to their classes. Treatments are collected in different sections as numerous as the classes of treatments prescribed under colored labels (Fig 31 A) whose colors coincide with those assigned by configuration to the classes and which are also visible in the "Filters" section (Fig 31 B);
- Category: to group treatments according to their categories;
- **Route**: to group treatments according to their routes.

4.00	and a second			
V. mall press				
B Wood and Social Second as	17-1 UNI			
	Regelier - Copies South Milling - Copies			
	Adjust - (Photocom Salaa Solar (40 mg - Inc) - (100:0022-015 MM			٠
•	REAMANA - Tunn IIIIII 1918 - Ula - Arti Ispania	д	-	
	Наблада — Сарция 1500 1271 на — Оса — сана а на до 750 на — «Растия» 1271 2211, Так на — сана 1321 2011, Так на	1	-	7
	Decision — VECANY TESTS, TEL Ng, W TEST-H at Y egyth — Roam of services sizes init againstate — Alconig, Thursday, Markeday, at TEST PA — unit "LINESTIN TESTS, TEL Ng, W TEST-H at Y egyth — Roam of services sizes init againstate — Alconig, Thursday, Markeday, at TEST PA — unit "LINESTIN			-
	indiana/au			
	Personance — Provide and information for constant for constant (2010): 20 will will be all fit with the information of a ferring to (2011) and the CONSTANT STATE (2010) (2011			100
				_
T. 2004				
at here				
at faile				
B. Seere .	Data Column Name			
and the second se				

Fig 32

In the example provided in Fig 32 the Grouping option "**Route**" is chosen. As a result, the prescribed treatments are grouped according to their route of administration into three different lists, indicated by the name of the route configured for each of them (also missing or unspecified route options are available).

4.13. Command Bar – Action buttons





THERAPY WEB PRESONITION

A command bar is available at the bottom part of the page (Fig 8 **C**), and it contains three buttons:

- **NEW** (Fig 33 **A**): to start the creation of a new prescription for the selected patient;
- **OVERVIEW** (Fig 33 **B**): to open a calendar time plan or a summary view;
- UPDATE (Fig 33 C): to open a therapy review window and update the treatment plan.

4.14. External resources (Help)

If the "Info" icon is present on the top-right corner of the "Summary" area of the prescription row (Fig 34 A), a link is available to configured external resources (an external URL or a file - the drug leaflet, for example).

		PAN-3-0,9 H at 2 mil										متعمم		b.							0	1	n.ene - 190	
								Zrig	10 ml	0.2 m	-	anan 2 mith	1946 P	441	56	NUMP BOKg								
1.444 Tue 17	OT ANA	52.697	13.44	54.438	25.464	06.458	117 204	08-844	00.498	10.404	11,000	1279	in Par	si m	63 PH	Di Pa	CLPM	oine	87 PM	DE PH	00 PM	10.74	11.94	
		energian SHORE	HL 300 H	ni																Heard 2007	-7 fy 424 SUSP			

Fig 34

Click the icon to access the external resource.

The resource accessed is the same triggered by the same icon on the Therapy Execution module (see paragraph 5.5.10).

If the "Info" icon is present on the top-left corner of the prescription window (Fig 35 **A**), a link is also available to configured external resources but this link is a specific prescription help, different from that described above (prescription instructions, modalities and contraindications, for example).

Gunity Daw				LADINA				Andre Carton	
internet () Speech Printy	Same while same	the design has	-	Any cold (London	Same.	Interest Parties	Andrew School Speech	And and Party	
			F	ig 35					

4.15. Create a New Prescription

To create a new prescription for the selected patient:

Click on NEW button (Fig 33 A).

A modal window is opened (Fig 36), and it contains two sections:

A Type to search.	Aettsfullmingt: strags
00 waters waters where werters	Anti-enetra
	// Bennoducegene arragentati
S-Aminolevulnic acid fing medicated planters	
5-Methoxypscralen	0 Desai-champirus
5-Methoxypstralen 70mg tablets	Inductive design
Abacavtr	Coat assessments
Abacavir + Laminudine	Attentioners dags
Abacavir + Laminudine + Dolutegravir	
Abacavir + Laminudine + Zidovudine	🥢 Manthe relationst reserval de
Abacavir 25mg/ml oral solution sugar free	Martin reissarra
Abacavir 300mg cablets	Openata
Abacavir 600mg / Laminutine 300mg tablets	Opinist amaginetists
Abacavir sulfate	(characteria
Abstacept	- Categori
absensive OS-multiwel extension for respectives may filled denote this denotion	(D)

Fig 36

• On the central section of the window, a search bar (Fig 36 **A**) on top of the page to search for the treatments to be prescribed.

Each row on the list of treatments refers to a specific treatment. The row can contain, if properly configured, a treatment description below the treatment name. A system option allows to display an additional search field making it possible to search also by treatment description.

Br	Name Tale Mineral	
	A Description Table to search T	
	60 event month attent states	
1 (1.3 DH EALL/RISEAN-3 C.PU	MICLONA .	
1.2 #ENDISTINZELIN CHEM	11/16	
1.2.5 78562566*6.56 PARTIES	194	
	Fig 37	

A DONE button (Fig 36 B) to close the window and five tabs are also provided:

- **All** (Fig 36 **C**) tab;
- **Moieties** tab (Fig 36 **D**);
- **Products** tab (Fig 36 **E**);
- Actions tab (Fig 36 F);
- **Mixtures** (Fig 36 **G**).

Each tab refers to a specific subset of prescribable treatments.

Select a tab to search for a treatment belonging to the subset indicated by the tab's name.

The **All** tab allows to search all the configured treatments.

The tab selected by default depends on a configuration option (*DefaultCatalogSearchLevel* System Option – See the document DSO ENG System Option for more information).

 On the right side of the window (Fig 36 H) filters are given to select treatments moieties, products, and actions - by configured class. Select one or more filters to display on the central page only treatments belonging to the selected classes. Click on Categories to select additional configured filters.

	Type to search	Arithdrorp: drap
5-Methoxypsoralen — Tatlet D056: 10 gram — Dral — 04/04/2025, 15:21	CO HORTEL PRODUCTS ACTORE HERVIRES	Anti-energy
	5-Aminolevulnic acid 8mg medicated plasters	💋 Bennskampine arragaraists
	5-Methoxypsoralen	🔴 Tenzi-Garapina
	5-Methoxypsoralen 20mg tablets	Industrie Rogs
	Abacave	Coal investitetics
	Abacavir + Lamivutine	O Mitoflarena ibap
	Abscaver + Lamoutine + Dolutegrave	- U. Muscle relatant revenuel drug
	Abacavit + Lamivudine + Zidovadine	2 · · · · · · · · · · · · · · · · · · ·
	Abacavir 20mg/ml oral solution sugar free	Muscle mhaarta
	Abscarir 200mg tablets	Openedia
	Abacavir 603mg / Lamiaudine 300mg tablets	Consid antiganters
	Abacave sufface	-
	Abatacept	Catagories
	Abatacept 125mg/1ml solution for injection pre-filled dispesable devices	The second second

Fig 38

 On the left, the current therapy for the patient can be shown clicking on the provided button to expand the section (Fig 36 I) and it contains the list of already prescribed treatments for the selected patient, and all the information on their forms and routes of infusions, dosages, and schedule (Fig 38 A).

If no therapy has been prescribed yet, the section will be blank.

ASTRODUCTURE AND VINDOS - DUDINOS RELAVIUMENTA RELATIVA DE SERVICE - DUDINOS AND ENTRE A VIDENTALIS RELATIVA DE SERVICE - DUDINOS AND ENTRE A VIDENTALIS RELATIVA DE SERVICE - DUDINOS AND ENTRE RELATIVA DE SERVICE - DUDINOS AND ENTRE RELA			CONDENT SHERING	1
ATTIVILIOUSA SALE IS OLANIAA INTEL VARAPEN- TADORISA INTEL APPORT ATTIVILIONE - CONTENSI INTEL VARAPEN- TADORISA INTEL INTEL CONTENCE - Professional INTEL CONTENCE - Professional INTEL - ethodose Advisor - Anni Specificat, Intel INTEL CONTENCE - Professional INTEL - ethodose Advisor - Anni Specificat, Intel INTEL CONTENCE - INTEL INTEL - CONTENCE INTEL INTEL INTEL - Ethodose Advisor - Anni Specificat, Intel INTEL CONTENCE - INTEL INTEL INTEL - Ethodose Advisor - Anni Specificat, Intel INTEL INTEL INTEL INTEL - Ethodose Advisor - Anni Specificat, Intel INTEL INTEL				
NUME VERSION ACTIVATION OF A CONTRACT OF A C				
Bold Wing - Good Scrift - and Links of the - status of the - stat				
	DALSE: WHE VE	a-DRA	UKDILUTTI	
- Serve a ray of 1712 - effective Autopa 2000 - unit special 2100 Autopa Aut	LEVERADOR	14.		
AND AND A CONTINUES				1.000 ·
Addition of Additional Addition of Additional Additi				
		-	- 1.001 mg	
		199	~ 1000 mg	
and the second s		-11-11	*	
		100		
		-		
			Fig 39	

If a therapy has already been prescribed and it is selected a second time (because it needs to be re-prescribed by emergency or administered in an alternative way or with a different scheduling), the history tab, containing the history of prescription and administration of the

therapy, will be displayed below the tab containing the current therapy. The possibility to prescribe the same therapy does not generate any warnings.

CO MURTER HODINITE ACTIONS	Arntzbuitningit drags.
DEXILUPROFENE	Anti-emetica
BURDENT	
BUPROFENE ISOBUTANOLANMONIO	1/2 Bertadampre singram
IBUPROFENE SALE DI ARGININA	Berro-chaingtrou
IBUPROPENE SALE DI LISINA	R Holester drugs
IBUPROFENE SALE DI NEGLUMINA	
BUPROFENE SODICO	Cocal assessments
BUPROFENE SODICO DIUDRATO	O Attractioneus druge
IBUPROFENE/CAFFEINA	// Manife refusert revenued dru
BUPROFENE/CODEINA	
IBUPROFENE/CODENA FOSFATO EMIDRATO	Muste reliaures
	🧭 Nansteradal ann àdhrenad
(C)	e
\checkmark	Other

4.15.1. New Prescription – Active Moieties Quantity/Dose step

Fig 40

To create a new prescription with a single active **Moiety**:

- Search for an active Moiety by entering its name (partially or in full) in the search bar (Fig 40 A) or use the class filters on the right (Fig 40 B). The resulting active moiety/ies are displayed in the central results section (Fig 40 C). If only Moieties are to be prescribed and we want to avoid retrieving Products and Actions, select the Moieties tab and perform the searches and filterings within it.
- Double-click on a treatment to be prescribed or select the treatment and then click on **Prescribe** button.
- > To abort the procedure click on **Done** button.

-DIGISTAT BI	LL. PATIENT & C Topic to Taxes Top 1	BC ST				-1.00 C
2) D	ship		BLERITCH	(card	- Westerland	
1			WARNING			
			4400		10	
-						
•						
					NAM.	G h. [111]

If allergies to certain drugs are listed in the patient's file and the source is correctly configured (the patient's file can be set as source file setting the **AllergySource** system option to 1), once the drug to be prescribed is chosen, the first screen displayed is the **Allergy** screen. A warning is displayed indicating that the patient is allergic to that medication and that a note must be entered to justify the choice. Entering the note is mandatory, otherwise the user cannot click the **Next** button - which remains disabled - and continue with the prescription.

Enter a note in the Allergy text field and click on Next button to complete the prescription.

15,7575		and a second		
🖌 Show all classes	6-days ALEYTICHA- das		36 * 4	8.6
O crans	SARDROSTERDONE DOLL 1 mg - AU22034, 10.29 AM		Active &	*
	(free)			
	9-TESTOGTERGHE 191010-9 mg - 4/12/034, 50/10 AM	6	ANT MAXAMED	*
	10PLID NOTE 2 mg - 32592554 at 435 PM - 500 PM - 3002024 at 200 AM	e		~
T Dead	ACTILISTEND - CAPADJ DULE BY MB - INTRAMUSCOLARE - 2020/2020 200 7M		-711.	
on Gaugeine	HUPROPERTY - COMPACENT DODE Rong - ORALE DALIFIC - 1/200324, DOE PW			*
ee douber				
er Samo				
be income				
THERAPY WEB	NEW OVERNEW PRINT	UPDA		

Thus, if a therapy containing an allergic medication is prescribed to the patient, a warning is displayed on the right of the central screen of the **Prescription** page. By hovering the cursor over it, the drug(s) to which the patient is allergic are displayed one above the other as in the example shown in Fig 42 **A**.

	MITSING ALLEINGES WIFGINGATION	
THERAPY WER PROCESSION		

Fig 43

Allergies listed in the patient's file can be ignored if the system option **AllergyMode** is set to 0 - False. If it is set to 1 - True, it is mandatory that possible allergies be listed, otherwise, when landing on the first prescription page of a treatment, the message Warning: "Missing Allergies Information" is retrieved (Fig 43).

30	Interaction	Giantty/Doee	httel/m	Review/Confirm
		WARNIN	1G	
		INVERTICALE OF CREWE AND A DECEMBER OF CREWE AND A DEC	ds with the current therapy	
	B ADER ACTUSALISEDE - SAGO	INERSONI IN CASHE SAW ASSESS WARRS ASSOCATE A PART IN CASE O	e ekonomis ka kanala wakonyi nilawene ika ka u	u attvite Patternija
		destinations (WITT	四
		Ć)		
				CANCEL NEXT
		Fig 44		(D)

In case one drug or treatment interacts with others, the first step of the prescription opens to the **Interaction** view (Fig 44): a warning is displayed (Fig 44 **A**) indicating that the chosen drug interacts with the current therapy. A configured section also appears in the center of the page (Fig 44 **B**) where further information about the risks and unwanted effects or possible effects of the interaction are provided. Then, before the quantity dose is selected, interaction notes can be entered in the text field provided (Fig 44 **C**). The note entry is not mandatory.

Click on **Next** button (Fig 44 **D**) to proceed to the next step.



Fig 45

The **Quantity/Dose** step of the prescription workflow is shown (Fig 45), and the name of the selected active moiety is shown on top of the page (Fig 45 **A**). The **DOSE** is *null* by default, the unit of measure is displayed next to the label, and the default route is also displayed (Fig 45 **B**).

Two dropdown menus are provided to choose, respectively:

- The active moiety Form (Fig 45 C);
- The active moiety **Route** of administration (Fig 45 **D**).

Variable number of cards is available in the central part of the screen (Fig 45 **E**) according to the configuration properties and parameters set for each treatment/drug mode. Different properties are associated with the different prescription modes, that represent specific clinical flows and configurations related to a set of catalog items.

20	interaction	Quantity/Dose				- ReviewCarlton
			IBUPROFENE SALE DI LIS	INA		
		n Bos With Durintern Bolo with Prolig	Bate any ProKg Solution	Solution Without Speed	Solution Without Volume	

Fig 46

There can be moieties and products with a single mode of prescription (and then their properties and configuration sets are displayed directly on the treatment specification window) or with **multiple modes of prescription**. In the latter case, before configuring the properties, it is necessary to choose from the prescription modes, provided as clickable buttons (Fig 46 **A**).



Any active moiety or product can be configured as a prescribable treatment and has its own default parameters and values. Therefore, the appearance and features of the treatment specification window depend on the prescribed treatment.



The treatment can be configured to enable only one possible prescription mode. If this is the case the selection buttons shown in Fig 46 are not displayed and the specification window (Fig 45) is instead directly displayed.

The button **Cancel** (Fig 45 **F**) is provided and enabled while the **Next** button (Fig 45 **G**) is disabled and will be enabled once the Quantity/Dose section is filled and completed.



Enter the desired values in the fields provided, either manually or by using the up and down arrows to increase or decrease the amount of value added by a unit.

When forms or routes are selected or the first values are entered into the available fields within the different cards, a blue **Change** button (Fig 47 **A**) appears. The **prescription script** (Fig 47 **B**) is filled in progressively with each choice and entry made.

Some configured fields are dependent on each other (e.g. Concentration, Amount and Volume or Volume, Speed and Duration). As a result, once a value is entered in two related fields, the corresponding value in the dependent field is automatically calculated.

By clicking **Change** button, the partial prescription string is reset, and the list of modes associated with the selected active moiety is displayed.



Choose a mode or, if only one mode is available, select a form from the provided dropdown menu (Fig 48 A);



- Select a route of administration from the provided dropdown menu (Fig 49 A);
- > Fill in the fields of all the available Quantity/Dose prescription properties cards;

Louis Capaire		•	Capsule			Farm Capsule		×
	Ато 500	unt.		Amour 1250	*		Атос 150	ent ÷
	mg	•		mg	*		mg	·•
	1 200 mg	B T		Fig 50	A me (D) T		1 200 mg 00	1020 mg T

For some parameters, such as **Amount** in the examples shown above, **lower** (Fig 50 **A**) and **upper** (Fig 50 **B**) **limits** are given, indicated with the exact dosages and specific symbols. When the value entered in the provided field is greater or less than the configured limits, the value of the exceeded limit is highlighted in yellow (Fig 50 **C**). When the value entered is correctly within the given limits, no highlighting occurs. Entering a dose outside the suggested limits does not stop the prescription creation process, but a note must be entered at the **Review/Confirmation** stage to justify prescribing values outside the configured range.



If an unexpected value is entered in some field, or the unit of measure is changed from the originally selected one (or from the default one), a **Verify** button (Fig 51 **A**) appears to

confirm the inserted value. In addition, a yellow warning (Fig 51 **B**) is displayed at the bottom right of the card containing the value to be verified. Moving the cursor over the icon, a tooltip (Fig 51 **C**) appears warning the user about the "unexpected input, verification of the value is required".

To verify or validate the inserted value:

> Just click on the **Verify** button. The warning icon disappears.

In particular, if a value has been entered with a specific unit of measure, if that unit of measure is changed, the value originally entered is recalculated and modified according to the new unit of measure (e.g., if the value entered is 1 and the unit of measure is changed from g to mg, the new value will be 1000).



Fig 52

It is also possible to enter the weight of the patient (Fig 52).

Click on **Next** button (Fig 47 **C**) to proceed to the **Schedule** step.



The procedure just described for creating a prescription from an active moiety can also be replicated for **Products** and **Actions**.

- Click on the "Products" or "Actions" Tab and search for the product by entering the name (partially or in full) in the search bar provided;
- > Double click on the desired product/action;
- > Proceed to configure and customize the treatment as previously described.
- > Click on **Next** button to proceed to the **Schedule** step.

Baaterby Doow					- shide	INTERNET	i.					
	SCLUDOAE METTO	oiv sola			here .		111					
	1	:	Ţ	:		Test Dire.	:		-	:		
	23	_	150				_	Ка				
				00040		-				~		
			0.25	10.05	-	150				C	O	
				G)							
												George -

Fig 54

Regarding **Products**, the Quantity/Dose section shows the parameters (Fig 54 **A**) to customize (their values may be present by default, but they can be edited) but also a **Components** section (Fig 54 **B**) with some concentration values already set by default (these values are not editable). They can also be sorted alphabetically by name by clicking on the **Name** button (Fig 54 **C**) or by decreasing concentration values, clicking on **Concentration** button (Fig 54 **D**).

The workflow required to prescribe mixtures is slightly different. See next paragraph (4.15.1.1) for the description.

4.15.1.1. New prescription – Mixtures

To prescribe a mixture.

Select the "Mixtures tab" (Fig 55 A).

All the configured mixtures will be listed on the left (Fig 55 B – in the figure only "Mixture test" is present). It is still possible to use the search functionalities and the filters on the right to find the required mixture.

CO HOTEL FOUND ACTION MARKED	Armuladowergie druge
	Arti-ersettisa
	💋 Recurskampies arragerists
	e Bernst-dausgeren
	behazian druga
	🛞 Loui avestitetus
	O Machinesen drops
	💋 Marsche entrannet etwernal strag
	🔴 Maschereiteaarte
	O Cynoids
	10 Optimil arragionists
	<u> </u>
	···· Cetagorae
	ETANKA KOUCH ENTITIER CO

Fig 55

Select the required mixture. The corresponding row will be highlighted (Fig 56 A).



Fig 56

Click the **Prescribe** button (Fig 56 **B**).

The screen listing the possible administration modes is displayed (Fig 57 A).

	Mixture Test	
6	Mixture Bola Mixture Ali	

Fig 57

Click the required administration mode.

The Quantity/Dose selection screen will open (Fig 58).

0	Arb.	DW/YE		5	olume		2	Concentration		
			•			•			•	
	mg		-	mi		-	mg/ml			30
				+ ADD	COMPONENT					
				CON	POMEN73					
			Acatta	×		Part	leit aater	×		
	0	Concentration			Enveprinatio		Volume			
	B	0,5	• •		8,0 milimi	•	4 ml	:		
		-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1					, a			

The default administration values are indicated on top (editable - Fig 58 A).

The components of the mixture and their values are indicated in the middle of the screen (Fig 58 \mathbf{B} , Fig 59).





The values of the components can be changed.

Either use the arrows placed on the right of each value to increase/decrease the Concentration/Amount (one unit per click) or select the value and then type the new Concentration/Amount to indicate the new value.

The components can be removed or added.

To remove a component:

Click the X button indicated in Fig 59 A.

User confirmation is required (Fig 60)
REMOVING PUR	IFIED WATER
Contir	ue?
CANCEL	ок

Click **OK** to remove.

To add a component

> Click the **Add Component** button indicated in Fig 59 **B**.

The following window, listing all the configured components, will be displayed.

Type to search	Anticholinergic drugs
5-Methoxypsoralen	Anti-erretics
Abacavir	💋 Benzodiazepine antagonists
Abacavir + Lamivudine	Benzo-diazépines
Abacavir + Lamivudine + Dolutegravir	Induction drugs
Abacavir sulfate	Local anienthetics
Abatacept	
Abciximab	Miscellaneous drugs
Abemaciclib	Muscle relaxant reversal drugs
	CANCEL BELLEC
Fig. 64	

Fig 61

Components can be either Moieties or Products. Click the tab corresponding to one or the other to display the required list (Fig 61 **A**).

The components are listed on the left (Fig 61 **B**). It is possible to use the available search tool (Fig 61 **C**) or the filters on the right (Fig 61 **D**) to find the required component.

Click the required component. The corresponding row will be highlighted (Fig 62 A).

1 CAN DE ACREMINATION	Anticholinergic drugs
MOJETIES PRODUCTS	Anti-emetics
-Methoxypsoralen	
ibacavir	Benzodiazepine antagonists
Abacavir + Lamivudine	Benzo-diazegines
Abacavir + Lamivudine + Dolutegravir	Induction drugs
bacavir sulfate	Local anaesthetics
ibatacept	
bciximab	Miscellaneous drugs
bemaciclib	🥢 Muscle relaxant reversal drug
	CANCEL SE

Fig 62

Click the Select button (Fig 62 B).

A window making it possible to indicate the component values will open (Fig 63).

ABCIXIMAE			
Concentratio	en		
mg/mg	C		
Amount Volu	ume		
		CANCEL	ADD COMPONENT
Fig 63			

- Select the appropriate terminology for the specific type of administration. That is: select "Amount" or "Volume" according to the specific administration type (Fig 63 A).
- If "Amount" is selected, the window changes as shown in Fig 64.

Concentration		Amou	NT.
1	:	0	
mg/mg		mg	
	O Amount	Volume	

If "Volume" is selected, the window changes as shown in Fig 65.

Concentration		Volun	
1	:	0	:
mg/mg	2.200	ml	
	Amount	O Volume	



- Insert the required values (Concentration, Amount/Volume).
- Select the appropriate unit of measure on the available drop-down menu (Fig 66 A).



When all the required data is specified (Fig 67 A),

Click the Add Component button (Fig 67 B).

Concentration	ABCXXM	Volume		
1	;	100	÷	
mg/mg	•	ml		
Ал	mount O	Volume		
				6



The new component will be added to the mixture and displayed on the Quantity/Dose selection screen (Fig 68 A).

Amount 20	tolar		Concentra 4	tion	
mg	* 5 mi	•	mg/ml		
A	+ ADD CON	IPONENT			
V	самися				
Abdenah Comanitation Volume	X Acres	io X	Concentration	ef auter Volume	×
0,5 🛟 100	0,5	10	0,8	4	:
nighting and	reging	mg	est/est	est	

Click the Next button to proceed to the Scheduling section, described in paragraph 4.15.3.



If the values specified for the new component are not coherent with the already existing values, the **Next** button is disabled and a specific warning is displayed.

4.15.2. New Prescription – Range specification for Quantity/Dose

Some treatments can be configured to allow the user to prescribe a range of Quantity/Doses instead of a specific value. That is, for example, prescribing a dose that is between 4 and 6 mg instead of prescribing 5 mg. The actual quantity will be decided at execution time, depending on patient's parameters and condition.

If, for a treatment, the "Range" option is enabled, the following screen is displayed after clicking on the **Prescribe** button (Fig 69).



Fig 69

The screen offers the possibility to prescribe using the "Range" option.

Click Yes to select the "Range" option (Fig 69 A).

The Quantity/Dose selection screen opens (Fig 70).



Fig 70

The parameters for which the "Range" option is enabled are characterized by the "From" and "To" fields, where the minimum and maximum range values can be specified. See for example Fig 70 **A** and Fig 71.





If the "From" field is filled with a value that is bigger than the one specified in the "To" field, then the "From" value is automatically brought back to the preceding logically possible value.

In the case of mixtures prescription, if a range is specified for the mixture, as in Fig 72 A, then the range is automatically calculated for the related components (Fig 72 B).

1.40.1.1.1.W				Contract of			
Speed	2.725	scentration .	0.57143	nition .			
7		•	0.5/145	•			
	mgimi		h				
14	:						
		Armusert					
mih	from 10.00	20					
	A 10.9	•					
	W 1.						
	21.8						
	mg						
	mg						
	mg			Electrony	20 Magaza (14 Magaza)		
	mg			+ 400 0	MPONENT		
	mg			and the second se	SMPONENT SN(NIS		
	mg			COMP	Contraction of the local division of the loc	ACCOUNT	
	mg	6.0 Concentrat	ion .	COMP	Contraction of the local division of the loc	Arr	einet (*
	mg	Concentrat	int .	eran Amount	Cancerne	atten An Fram	ours in
	mg		ion .	cour	ONENTS	Arr	
	mg	Concentrat	int .	cover Amount (X)	Cancerne	atten An Fram	olett (***
	ng	Concentrat	int .	erati Amount	Cancerni 0.5	atten An Fram	

Fig 72

After Quantity/Dose specification:

Click the Next button to proceed to the Scheduling section, described in paragraph 4.15.3.

4.15.3. New Prescription – Scheduling step

200	carewytow	Terester	- Texture and a
		ADCHORUS ROMTERINGSFSE — SOLUDINE WETTPOLYSISY DODE: 200 Milat 100 mBh — INTRAMUSCULARE DUUMD —	
		investor canonical bats when science castor	(D.Everymon
		ß	۵
			GANCEL

Fig 73

The page lands on the **Schedule** page and the blue **Back** button (Fig 73 **A**) is provided to go back to the previous step.

The list of six scheduling options presented as individual buttons is displayed (Fig 73 **B**). The following options are provided:

- Immediate: to immediately administer the prescribed treatment;
- Conditional: to administer the prescribed treatment only if certain conditions occur;
- **Daily**: to administer the prescribed treatment in one day, one or more times, at given times;
- **Weekly**: to administer the prescribed treatment on different days during a week, at given times;
- Schema: to administer treatment according to a fixed, user-defined pattern;
- **Custom**: to administer treatment according to a fully customizable schedule.
- Emergency: to administer an emergency executed treatment.

It is possible to set the automatic plan type selection (conditional, weekly, daily, schema etc...) by configuration. Once a treatment is selected, the corresponding window will automatically be displayed.

4.15.3.1. Schedule - Immediate





If the **Immediate** schedule is chosen, the treatment is immediately administered. The message "Single administration and immediate at" followed by the administration time (the current time when the **Immediate** scheduling option was selected) is displayed (Fig 74 **A**). No further configuration is required.



It is possible to only set the **tolerance time** i.e., the time interval before and after the specified administration time - in the case of prescriptions with immediate schedules – within which the administration is considered "on time", by choosing one of the options

available in the dropdown menu provided (Fig 75 **A**). If a tolerance of 15 minutes is specified for an administration prescribed for 11:00 a.m., the administration is on time if it is performed between 10:45 a.m. and 11:15 a.m.

The following options are given:

- 00:05
- 00:10
- 00:15 (the default tolerance value is set to 00:15 minutes)
- 00:20
- 00:30
- 00:45
- 01:00
- 01:30
- 02:00
- Set the tolerance time.

The blue button **Change** (Fig 75 **B**) is enabled, and it is possible to change the selected scheduling, going back to the selection window shown on Fig 73, to opt for another schedule.

The blue **Next** button (Fig 75 **C**) is enabled, and it is possible to proceed to the **Review/Confirmation** stage.

4.15.3.2. Schedule - Conditional

	QuentyOne	Schwoure	Researcheters
		illugeration Copyoin 2052: 500 mg Chat It the condition A happens, then presente	8
		concitiones	
	ation A happens, then prescribe X		
	20127		
B	O Roser		

Fig 76

If the conditional administration is chosen, the prescribed treatment will be administered only if certain conditions, that must be specified in the textbox provided, are met.

- Set the condition or conditions to be met in the textbox available (Fig 76 A);
- Choose between the Once or Repeat options (Fig 76 B), selecting the corresponding radio button to indicate whether the treatment is to be administered only once or whether, after being administered the first time, it is to be retained in the treatment plan in order to be repeated in the future.

The red button **Change** (Fig 75 **B**) is enabled, and it is possible to change the selected scheduling, going back to the selection window shown on Fig 73, to opt for another schedule.

The blue **Next** button (Fig 75 **C**) is enabled, and it is possible to proceed to the **Review/Confirmation** stage.

4.15.3.3. Schedule – Daily

			n - Capsule reg - CYV -		
		D.	66.V.		
	1/30/2023 1 00 PM	6			
	1130/2023 1 00 PW .			0	
14	000	(THM)		V I	
	0.15	+		- O Repart	
~			1.27		
	100.004	200 PM	3300 884	#00 PM	
	2000,049	8477.356	3.000 PHd	8.02.94	
	0.00 PW	10000 PM	THEFT	12:DE WM ⁺⁺	
	D It of Ald	2:00.AM**	331.444	A 55 A 54"	
	2.00 AM	0.03.864++	2102 ANI	automi-	
	8:02.AM 71	10:00 AM 7	FEXE AM 1	12:00 PM **	

If daily administration is chosen, the prescribed treatment can be administered once or several times a day at set times. To schedule the treatment administration times, multiple fields are provided:

- The **Starting from** field (Fig 77 **A**) to set the date and time of the first administration of the prescribed treatment;
- The **tolerance** field (Fig 77 **B**), to set the tolerance time (the default value is 00:15 minutes);
- Once and Repeat radio buttons (Fig 77 C);
- A card grid with **24 full-hour-buttons** to be selected (Fig 77 **D**). The 24 buttons correspond to the 24 hours of a day. The first hour is the one set in **Starting from** field or, if the set hour is not a full hour, the first hour will be the nearest full hour. The indication "+1" means that those hours are relative to the next day.



Fig 78

It is possible to set and configure the timetable on 24 hours as well as 12 hours am/pm (Fig 78 **A**). To prescribe a daily schedule:



- Click on the field itself or on the calendar button (Fig 79 A) provided in the Starting from field to open a dropdown calendar view;
- Select the date in the Date tab (Fig 79 B) and the time in the Time tab (Fig 79 C), then click on Next button (Fig 79 D). It is possible to specify the start of treatment on a future day and/or at a future time, but not in the past;



Fig 80

Set the tolerance time and select the desired options from the available radio buttons. If repeated prescription is selected, then the End date checkbox (Fig 80 A) appears.



Checking this box, it is possible to set the end date for the daily treatment (which can then be administered on multiple days according to the fixed times set) on a calendar view (Fig 81 A);



Select the hours on the hour-cards grid by simply clicking on them (double-click the selected hours to deselect them). The selected hours are colored in blue (Fig 82 A).

The red button **Change** (Fig 75 **B**) is enabled, and it is possible to change the selected scheduling, going back to the selection window shown on Fig 73, to opt for another schedule. The blue **Next** button (Fig 75 **C**) is enabled, and it is possible to proceed to the **Review/Confirmation** stage.

4.15.3.4. Schedule – Weekly





If **weekly** administration is chosen, the prescribed treatment can be administered once or several times a day at set times for one week. To schedule the treatment administration times, multiple fields are provided and most of them have already been described for the **daily** prescription plan. The previously described functionalities are integrated with a **button/card grid with the 7 weekdays** (Fig 83 **A**) to be selected. The 7 buttons correspond to the 7 days of a week. The first day is the weekday of the one set in **Starting from** field. To prescribe a weekly schedule:

- Click on the field itself or on the calendar button provided in the Starting from field to open a dropdown calendar view;
- Select the date in the Date tab and the time in the Time tab, then click on Next button;
- Set the tolerance time and select the desired options from the available radio buttons. If repeated prescription is selected, then the End date checkbox appears;
- Checking this box, it is possible to set the end date for the weekly treatment (which can then be administered on multiple weeks according to the fixed days and times set) on a calendar view;
- Select the hours on the hour-cards grid by simply clicking on them;

Thursday	Friday	Saturday	Sunday
Monday	Tuesday	Wednesday	

Select the days on the day-cards grid (Fig 84) clicking on them (double-click the selected days to deselect them). The selected days are colored in blue.

When prescribing a weekly treatment plan, it is necessary to specify both the times of administration and the days of administration. The red button **Change** (Fig 75 **B**) is enabled, and it is possible to change the selected scheduling, going back to the selection window shown on Fig 73, to opt for another schedule.

The blue **Next** button (Fig 75 **C**) is enabled, and it is possible to proceed to the **Review/Confirmation** stage.

4.15.3.5. Schedule – Schema

	Buprofer	n Gosale mg Oral	
	DUTLE SAU	mg - une -	
	504	ENA;	
thereighter.			
11/30/2023 5:00 PM	22		
Terminer			
00.15	20	😑 Crest 🔘 Repeat	
Norther of ea	atulizità illivioi	besien everytene Dureken	



If schema administration is chosen, the prescribed treatment can be administered according to a fixed, generic pattern, like "Execute this treatment N times in X days". The system calculates the appropriate time intervals and enters administration orders from the time indicated in the **Starting From** field. To schedule the treatment administration times, multiple fields are provided and most of them have already been described for the **daily** prescription plan.

To prescribe a schema schedule:

- Click on the field itself or on the calendar button provided in the Starting from field to open a dropdown calendar view;
- Select the date in the Date tab and the time in the Time tab, then click on Next button;
- Set the tolerance time and select the desired options from the available radio buttons. If repeated prescription is selected, then the End date checkbox appears;
- Checking this box, it is possible to set the end date for the schema treatment on a calendar view;

	Ø	00.15	Crise O Inspect Sumble (Francisky) universe between executions Caratium
D	Franzis	:	
			Number (B) Strend Determination: Duration
€	At annevator	:	Hom •
			Martine of Executions - Executed Services Control & Control
Ð	97 (C)	:	Huars •

Fig 86

To define a schema,

- Use the buttons shown in Fig 86 (A, B or C) to select the appropriate option for the desired schema to be composed. Three buttons are available:
 - **Number of executions**, to specify the number of administrations. If selected, the **Execute** field (Fig 86 **D**) is provided to enter the desired number of administrations;
 - Interval between executions, to specify the time interval between one administration and the next. If selected, the **At interval of** field (Fig 86 **E**) is provided to enter the desired time interval. The unit of measurement (Minutes, Hours or Days) can be chosen from a dropdown menu provided;
 - Duration, to specify the period of time (how long) within which to perform therapy administrations. If selected, the In field (Fig 86 F) is provided to enter the desired period of time. The unit of measurement (Minutes, Hours or Days) can be chosen from the dropdown menu provided (Fig 87 A);

Quartity/Ocar		Schedule	ResearCartin
	0010.120 mg - 0	Bugrafiet — Capule Nal — Statestik Jack — unit (2012)223, 830 AM	
		30+5104	
	Samingham.		
	11/20/2023 5:90 PM	3	
	Telesce.		
	00.15	* O head	
	S Number of Station	Inter Contractioner Lawrahow Contra	
lauin.			
5			
E-marked of			
8	Hours	*	
÷.,	Minutes		
24	A man		
	Days		
		Fig 87	

Two buttons must be selected at a time in order to establish a pattern. If **Number of executions** and **Interval between executions** are selected, like in the example shown in Fig 87, values can be entered in the **Execute** and **At interval of** fields. The **In** field is populated accordingly, since the system automatically calculates the total time period by multiplying the set number of administrations by the chosen time interval.

	1	• • • • • • • • • • • • • • • • • • •
	C Testforr III too	
3	:	
21	: Hours	*
10,33553333	Hours	

Fig 88

Similarly, if the selected fields are **Number of executions** and **Duration**, the user can enter the desired values in the **Execution** and **In** fields, and the value in the **At the interval of** field is automatically calculated.

	Toleana 00:15		• 😑 onca (August C
		Number of executions	Siderval hetween overcutions	
At Intervel of 8.25	:	Hours		
- 33 former 4	:	Hours		
A diamate				

Fig 89

Finally, if the selected fields are **Interval between executions** and **Duration**, the user can enter the desired values in the **At interval of** and **In** fields and the number of executions is automatically calculated.

- Choose the schema, selecting the desired fields to be filled in;
- Enter the desired values in the fields provided, either manually or by using the up and down arrows to increase or decrease the amount of value.

The red button **Change** (Fig 75 **B**) is enabled, and it is possible to change the selected scheduling, going back to the selection window shown on Fig 73, to opt for another schedule.

The blue **Next** button (Fig 75 **C**) is enabled, and it is possible to proceed to the **Review/Confirmation** stage.

4.15.3.6. Schedule – Custom



If custom administration is chosen, the prescribed treatment can be administered according to a fully customizable schedule. In this case, the orders that will be generated are all explicitly specified, one by one. To prescribe a custom treatment:

Set the **tolerance** choosing one option from those provided in the provided dropdown menu;



Fig 91

- Click on + Add Event blue button (Fig 90 A) to set the schedule of the first "event" from the provided calendar;
- Set the date in the **Date** tab and the time in the **Time** tab, then click on **Next** button;



Enter all desired event orders by clicking the button and completing the customization of the schedule for each administration;

For each order, a card with the date and time of administration of the customized treatment is displayed (Fig 92 **A**). The events are displayed and listed in chronological order (even if they've been created in non-chronological order). To cancel an incorrect or no longer needed order, a **bin** button is available (Fig 92 **B**).

To cancel the order,

> Just click on the **bin** button and confirm the deletion.

The red button **Change** (Fig 75 **B**) is enabled, and it is possible to change the selected scheduling, going back to the selection window shown on Fig 73, to opt for another schedule.

The blue **Next** button (Fig 75 **C**) is enabled, and it is possible to proceed to the **Review/Confirmation** stage.

4.15.3.7. Schedule – Emergency

	Quantiy Dose	- Indu	Append Cardon
		ACCETELENCE FOR HALE DE CLANINA TETELE, TRE NIGHT - 1004 (2004, 10:00)	
		Evel Aug Aug In	
		1440 1440 1140 1240	
		Factor Factor Factor Textor	
DIVOR		(Danka)	GANCEL HERE
and the second s		Fig 93	

In case the emergency option is chosen, only one administration with the same or previous time can be prescribed. In the example provided in Fig 93, the prescription was registered at 16:45. Eight previous full hours are selectable.

- Select an hour from the clickable ones provided in the grid and then
- click on the **Next** button.



The number of selectable hours in the past is configurable, depending on the EmergencyHoursLimit System Option. See the document DSO ENG System Options for more information.

4.15.4. New Prescription – Review/Confirm step



Fig 94

After clicking the **Next** button, the user can review and complete the prescription in the last available window, under the **Review/Confirmation** tab, where the summary of the prescription can be read.

Three sections are provided:

- The full prescription string (Fig 94 **B**) is displayed on top of the page, with the set dosages, unit of measures, form and route of administration and the full scheduling plan for the prescription. The prescription field is not editable;
- A summary section (Fig 94 **C**) shows the prescribed values with their unit of measure. The summary is also not editable;
- An additional field (Fig 94 **D**) is also available to enter a **note** to the prescription. It is not necessary to fill in this field, but to do so:
 - Manually enter the note inside the textbox provided.

To save the prescription:

Click the blue Save button (Fig 94 A).

Name of State			
Marchine - Gayane Mala Marceg, - pro - materials (marce regulation and regulation and reg			*
Augente - (Parcessen halfe) Biller Helleng - Scale - (11000000 h.11494			
Ryangene – Sanat Mille Stang – Kan Asan Agama	5	-	
Миникана - Старили Миника в ИСтину - Стари - илика и Лиу и Пол Ана - иликана - 10110000, Пол ана - илик 101000000, Тол ана	12	-	*
Fig 95			

The prescription is added to the current therapy of the selected patient on the Prescription main page (Fig 95).

4.16. Prescription with double signature

A treatment can be configured to require a double signature at prescription time and/or at administration time. For prescriptions, it is necessary that a second user, different from the one that prescribed the treatment, validates the prescription. The prescription procedure is the same described in paragraph 4.15 but, if a double signature is required for the prescription, a specific icon - \bigcirc - is displayed on the corresponding row on the prescriptions table (Fig 96 **A**).



For the signature procedure at execution time see paragraph 5.7.

Two different procedures can be configured to allow a second user to sign the prescription, depending on the *PrescriptionSignMode* system option.

If *PrescriptionSignMode* = 0 the signature must be performed on another Digistat session. That means that the prescription must be signed by another user logged on a different Digistat session.

If *PrescriptionSignMode* = 1 the signature must be performed within the same Digistat session, where the credentials of a different user are requested.

Both cases are described below.

4.16.1. Signature on a different Digistat session

In Fig 98 the Abacavir treatment is prescribed. In the current configuration, Abacavir requires a double signature (Fig 98 **A**).

		cinet		
Abacavir — Tablet DOSE: 10 mg — Oral — 09/04/2025, 10:42			A> 8	ACTIVE 🗸
)
NEW	OVERVIEW	UPDATE Fig 98		

Fig 98

If the *PrescriptionSignMode* system option is set to 0, the **SIGN** button on the command bar is disabled (Fig 98 **B**).

To sign the prescription, it is necessary to log in with different credentials on a different Digistat session and select the same patient.

The **SIGN** button will be enabled for the second user. See Fig 99 **A**.

header I half DOCH TO MENU UPDATE Fig 99 P Click the SIGN button (Fig 99 A). the following window opens (Fig 100).			OTHER	÷			
Fig 99 • Click the SIGN button (Fig 99 A). the following window opens (Fig 100). • Concert • Co				e		10.000	
Fig 99 Click the SIGN button (Fig 99 A). the following window opens (Fig 100). Concert							
Fig 99 Click the SIGN button (Fig 99 A). the following window opens (Fig 100).							
Fig 99 > Click the SIGN button (Fig 99 A). the following window opens (Fig 100). Accurr - Tablet tops: 10 mg - Oral - 09/04/2025, 10:42 CANCEL							
Fig 99 Click the SIGN button (Fig 99 A). the following window opens (Fig 100).							
Fig 99 • Click the SIGN button (Fig 99 A). the following window opens (Fig 100). • • • • • • • • • • • • • • • • • • •					0		
Click the SIGN button (Fig 99 A). the following window opens (Fig 100). Image: Tablet Image: Tablet <td>n</td> <td>OVERVIEW</td> <td>UPDATE</td> <td></td> <td>SIGN</td> <td></td> <td></td>	n	OVERVIEW	UPDATE		SIGN		
the following window opens (Fig 100).							
Abscavir - Tablet Image: Crail - 09/04/2025, 10.42 Abscavir - Tablet Image: Crail - 09/04/2025, 10.42	Click the SIGN	button (Fig 997	A).				
Atland Post 10 mg - Oral - 09/04/2025, 10:42		pope (Fig 100)					
Abacavir Tablet PDSE: 10 mg Oral 09/04/2025, 10.42	e following window c	pens (ng 100).					
CANCEL	e following window c		ou are authenticated as			ELECTA	41
	ıbacavir — Tablet		ou are authenticated as Marca	08	C.	SELECTA	a
	ıbacavir — Tablet		ou are authenticated as	08	ACINE	ELECTA	44
	ıbacavir — Tablet		ou are authenticated as Manua	08	ACINE	ELECTA	41
	ıbacavir — Tablet		ou are authenticated as Manua	08	ACINE	SELECT A	4.1
	ıbacavir — Tablet		ou are authenticated as Manua	08	ACINE	SELECT A	4.1
	ıbacavir — Tablet		ou are authenticated as Manua	08	ACINE	SELECT Α	4.1 ;
	ıbacavir — Tablet		ou are authenticated as Manua	08	ACINE	SELECT A	44
	ıbacavir — Tablet		ou are authenticated as Manua	08	ACINE	SELECT A	
	ıbacavir — Tablet		ou are authenticated as	08	ACINE	SELECT A	
	ıbacavir — Tablet		ou are authenticated as	08	ACINE	ELECTA	
	ıbacavir — Tablet		ou are authenticated as	08	ACINE	SELECT A	
	ıbacavir — Tablet		ou are authenticated as	08	ACINE	ELECTA	41
	ıbacavir — Tablet		ou are authenticated as Manua	08	ACINE &	~	

The window lists all the prescriptions requiring a signature. In the figure only the Abacavir prescription is present (Fig 100 A).

The user currently logged is indicated on top (Fig 100 B).

The Select All button (Fig 100 C) allows you to select all the items listed on the window.

Check the button placed on the right of the rows corresponding to the items to be selected (Fig 101 A).

The SIGN button will be this way enabled (Fig 101 A).



Click the SIGN button (Fig 101 B).

The selected prescriptions will be this way double-signed.

4.16.2. Signature on the same Digistat session

In Fig 102 the Abacavir treatment is prescribed. In the current configuration, Abacavir requires a double signature (Fig 102 **A**).



If the *PrescriptionSignMode* system option is set to 1, the **SIGN** button on the command bar is enabled (Fig 102 **B**).

Click the SIGN button (Fig 102 B).

The following window opens (Fig 103).

You are authenticated as and particular to the second s	
(A) WARNING	
You have no prescriptions to sign	
This happens because all the prescriptions have been created by	an a
	CANCEL

Fig 103

The window warns the logged user that the prescriptions requiring a double signature must be signed by a different user (Fig 103).

To insert the credentials of a different user

Click the icon indicated in Fig 103 B.

The screen changes in the following way (Fig 104).

Usemame	
Password	

BACK		CANCEL 1999
	Fig 104	

Insert the credentials of the second user in the "Username" and "Password" fields (Fig 104 A).

The **Authenticate** button will be enabled (Fig 105).

	AUTHENTICATE	
	Authenticate yourself in order to sign your prescription orders in the following step	
	Mathematica	
	+	
	AJTHENTICATE	
BACK		CANCEL DON
	Fig 105	
\succ	Click the Authenticate button (Fig 105 A).	

The following window opens (Fig 106).





The window lists all the prescriptions requiring a signature. In the figure only the Abacavir prescription is present (Fig 106 A).

The user currently logged is indicated on top (Fig 106 B).

The Select All button (Fig 106 C) allows to select all the items listed on the window.

Check the button placed on the right of the rows corresponding to the items to be selected (Fig 107 A).

The **SIGN** button will be enabled (Fig 107 **B**).

	2. You are authenticated as		DESELECT ALL
Abacavir — Tablet DOSE: 10 mg — Oral — 09/04/2025, 10:42		08	



Fig 107

Click the SIGN button (Fig 107 B).

The selected prescriptions will be this way double-signed.

4.17. Overview the created Prescriptions



The second button provided on the command bar is the **Overview** one (Fig 108 **A**). Clicking on it, a window opens containing a **summary** of all the active prescribed treatments and the **statuses** of the corresponding orders on a calendar view.

aprile 2024 4 10240 P	TBUEFLAN	
the last set on two and the		[int] as
4 3 4 2 4 1 8 6 7 1 1 1 1 11 6 7 1 1 1 1 1 12 23 34 22 34 27 34 0URICATINE 20 35 4 24 34 27 34 0URICATINE		inin Tapala Ini ini ita ita

Fig 109

The page opens to the TimePlan tab (Fig 109 A).

Filters related to the classes of drugs prescribed in the therapy are displayed on the left (Fig 109 **A**). By clicking on one or more button filters, only the timeplan related to the prescription drugs belonging to that class will be displayed (Fig 109 **D**).

A calendar (Fig 109 **C**) is provided on the left part of the page. The current date is highlighted, but it is possible to scroll through the calendar to select days in the past and future directly by clicking on them. As a day is selected, the corresponding time plan (Fig 109 **D**) for that day is displayed in the central part of the screen.

4.17.1. Overview – Timeplan view



In the **timeplan** view, the schedule of administered orders or to be administered orders for each prescribed treatment is shown in a grid. Treatments are listed as individual rows in the grid (Fig 110 **A**). A **red vertical bar** (Fig 110 **B**) marks the "now" moment, and the orders scheduled for each treatment are represented as **cells** (Fig 110 **C**) that can display different colors:

- Gray, if it is an order to be administered in the future;
- **Green**, if it is a "ready" order, an order to be executed at the present time, i.e., within the time frame configured as the "tolerance interval";
- **Red**, in case of an unexecuted and overdue order;
- **Black**, in case of a correctly executed order;
- **Cyan**, in case of running administrations.

Durative administrations are displayed as a cyan bar connecting a starting point and an end point. See Fig 111 for an example.



It is possible to change the time plan display using the three buttons available at the bottom left part of the Overview page. Those same buttons are also provided in case the **Summary** view is selected and they are:

• Weekly button (Fig 110 D), to show the entire week from Sunday to Saturday (days when no orders are prescribed are also shown). Each day is divided into 4 time slots (00-06, 06-12, 12-18, 18-00), represented as separate cells (Fig 110 G) within which the

orders are represented as colored portions of the cell placed at the corresponding time;

100000-000 - 4 - 7000 - 4		B	1000	
	Ages Ages			 1
•	Re nore	B		
100 (M) 11				
		Fig 112		

• **Daily** button (Fig 110 **E**), to show the full day in detail: the day is shown at the top left of the grid (Fig 112 **A**), while 24 cells are provided (Fig 112 **B**), one for each full hour;

	CA-	 14	100.0			- 24	122
the second second second	August .	 	-		-		- 61
	from succession		-				
			1				
		V		1			
THE LET LET		4. 10/22					

Shift button (Fig 110 F), to show administered or to be administered orders within an 8-hour shift. The day is shown at the top left of the grid (Fig 113 A), while 8 cells are provided (Fig 113 B), one for each full hour of the displayed shift.

Two arrows (left and right arrows) are provided in the command bar, to move between **time slots** and **days** of the week, **hours** of the day, and **shifts** (they can be configured, in the example the programmed shifts are as follows: 14-22, 22-06, 06-14), respectively. Finally, a **Close** button (Fig 110 **H**) is provided to exit the Overview page.

4.17.2. Overview – Summary view

Click the "Summary" tab (Fig 114 **A**) to display the **Summary** view (Fig 114).

higlio 202	£ - 3	8	TODAY	. 2	TIMEPLAN SUN	ARV B
11/10 (1994)	ant (pr	5909	1.376	1004	O Alconnati O Abriance	
-	2 3	- 4	6	6	ABCIXIMAB	
0	1.7	्य	Ω.	E		
14. 15	10 5	7 18	. 10	S	ABIRATERONE	ن ب
21 22	20 2	25	20,	37		
28 29	31 3	E				
	6					
	¢	1				
WEEKLY	C DN	er i	SHIFT	r	6 B	cro

The Summary view displays all the treatments administered in a given time span. Two modes are available. Use the buttons indicated in Fig 114 $\bf B$ to switch from one mode to another.

1 - Aggregated mode (left button - 🛄 👛).

In aggregated mode the amounts for the same treatment are considered all together, as global amounts. Different prescriptions of the same treatment are calculated all together.

- Select the period using the options available on the bottom-left corner (Fig 114 C Weekly/Daily/Shift). The selected period is highlighted on the calendar (Fig 114 D).
- Click the row corresponding to a treatment to display the administration values in the specified period (Fig 114 E).

The following view is displayed (Fig 115).

5	TURNE	2				TIMEPLAN	SUMWARY			08	m
B	110	DDM .				O Abdanati O Abra	anne Ø Alace				
V	1	-	ABCIXIMAB							3	,
		27 27	ABIRATERONE			0				~	
21						2 13	3 gram				
		<u>()</u>	turi 7 baştır 2025	mar B	mar 8	gis 30	vien 11	sab 12	dom tå		
			ACACIA								,
	-	4			1.02					a	
	34 2	B 1 100 17 16 17 24 25 28 21	0 1 0 1 1 1 1 1 2 2 2 2 2 2	B I II II ABCOUMAB 10 17 10 12 II 17 16 10 21 ABCOUMAB 26 25 26 27 II 27 10 27 III ABRATEBONE 28 25 26 27 III 26 27 27 III IIII 28 25 26 27 IIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII	ABRATERONE ABRATERONE	B No DOM No No No Se 25 25 27 Se No No MBRATEBONE No No Mol 3 Mar 3 mar 9 Mol 2000 Mar 9 mar 9	B to two DDM © Meckenell O Me	B No D No No <td>Bit rive DDM ® decennal © About rest 00 rive rive ABCAIMAB 17 rive rive ABRATEROME 20 rive rive Open 20 rive Methoda 20 rive Methoda 13 gram ABRATEROME Additional Methoda ABRATEROME Down ABRATEROME Open ABRATEROME Down ABRATEROME Open ABRATEROME Down ABRATEROME Down 2 13 gram ABRATEROME ABRATEROME ABRATEROME Methoda ABRATEROME Down ABRATEROME Sab 12 ABRATEROME ABRATEROME</td> <td>Bit Med DOM Machine Dom 10 1 10 10 17 10 26 20 2 13 gram</td> <td>Bits res DOM Bits res DOM<</td>	Bit rive DDM ® decennal © About rest 00 rive rive ABCAIMAB 17 rive rive ABRATEROME 20 rive rive Open 20 rive Methoda 20 rive Methoda 13 gram ABRATEROME Additional Methoda ABRATEROME Down ABRATEROME Open ABRATEROME Down ABRATEROME Open ABRATEROME Down ABRATEROME Down 2 13 gram ABRATEROME ABRATEROME ABRATEROME Methoda ABRATEROME Down ABRATEROME Sab 12 ABRATEROME ABRATEROME	Bit Med DOM Machine Dom 10 1 10 10 17 10 26 20 2 13 gram	Bits res DOM Bits res DOM<

The timeline indicated in Fig 115 **A** shows, for the specific treatment, all the orders in time span selected on the calendar (Fig 115 **B**). The red bar corresponds to the present moment. The orders are placed on the timeline in the position corresponding to their time of administration (scheduled time for future orders, actual time for past orders). They are colored according to the general color code within "Digistat Therapy Web" (see above, paragraph 4.17.1).

The squares indicated in Fig 115 **C** and enlarged in Fig 116 display in an aggregate way the number of administrations of the same treatment and the administered volumes. "Aggregate" means that different prescriptions of the same treatment are here considered all together.

BAGAGLARYSE'S KHURPEN 1997							
	104	HALME	RANNEVIDAE.	SHOLD	10.00		12.148
	3	9 ml	0.77 ml	4	12 ml	9	27 ml
	-		11				

The squares shown in Fig 116 can refer to

- Total administerd orders and administered volume
- Administerd volume of (possible) current ongoing administration.
- Cancelled orders and corresponding total amount.
- Planned orders and corresponding total amount.

1 - Single prescription mode (right button -

In "Single prescription mode" the different prescriptions are separated, and the displayed amounts refer to a single prescription (Fig 117 **A**).

DOSE:	aterone Talsk	1 - 7 mi/h — Introvenc	ous — 07/07/2025, 10-49	 1	Aveau 3 gram		- 8	ADM tony GMMATR W	•
Abirat 0058:			1.00					annaithe an	^
A									
lun lug	un 7 1gila 2025	mar 8	rrar 9	go 10	ven 11	sab 1	2 do	m 13	
			2.09				e	ACOM tang	~
			not applicable — 07/07/	2025, 11:14				лти 107	~
FT			4	3					CLOS
F.	Abirz DOSI Acad DOSI	Abiraterone Tabi ODGE: 10 gram O Acaeta Not applic DOGE: 20 gram R	Abiraterone — Tablet DDSE: 10 gram — Oral — 07/07/2025 at 1 Acaela — Not applicable DDSE: 20 gram — Boute of administration	Abiraterone Tablet DDSE: 10 gram Oral 07/07/2025 at 12:00 Acada Not applicable DDSE: 20 gram Route of administration not applicable 07/07/ C	Abiraterone Tablet DDSE: 10 gram Oral 07/07/2025 at 12:09 Acaetia Not applicative DDSE: 20 gram Boute of administration not applicable 07/07/2025, 11:14 T	Abiraterone Tablet DDSE: 10 gram Ond 07/07/2025 at 12:00 Acada Not applicable DDSE: 20 gram Boute of administration not applicable 07/07/2025, 11:14 T	Abiraterone Tablet ODSE: 10 gram Oral 07/07/2025 at 12:09 Acaela Not applicable DOSE: 20 gram Boute of administration not applicable 07/07/2025, 11:14	Abiraterone - Tablet DDSE: 10 gram Oral 07/07/2025 at 12:00 Acada Not applicable DDSE: 20 gram Route of administration not applicable 07/07/2025, 11:14 C >	Abiraterone Tablet COSE: 10 gram Oral 07/07/2025 at 12:00 Cose Cose Cose Cose Cose Cose Cose Cose

The screen layout and working procedures are the same described for the aggregated mode (see above).

4.18. Update the Therapy Cycle



The treatment plan update procedure allows to update the prescriptions' orders and to delete the expired orders or the orders to be removed. To update the treatment plan:

Click the **Update** button on the command bar (Fig 118 **A**).

PRESCUTION DATE	THE RANK RECEIVE			Y
	epida mito Per la econo esti Ale	(1020028, 1130 PM	Area .	I
Analysis Analysis and a state of the state o		FINENER FIEPP	Algori	
Methoday - Sociale Million 8.003 mg - Cold - anna 4 Bar et 700 AM - effective, 1211 (202)	1708 AM 1899 12/31 (2023-710) AM	CONTRACTOR AND	RAN	Ī
Glastine – trifactori Glastine 100 mg in 1000 ml at 1 mg/h – Booke of admonstration not appl Marving, Phonesis, Tablaka, WebPenday at 1000 PM – onto 12,00000		CORDING CORD PRO	naz <mark>(</mark>	>
And second - Roader and labor for statistic for species from 20 of 4018 with - Schulerman - 2 towards 345 - and 1271	C1121 (#100.444	FURNISHED BOOM	F	B
		CERT COSCER BOOK AND	here .	Π

Fig 119

The **Therapy Review** window opens on the **Prescription** tab (Fig 119 **A**) showing all the created prescriptions' orders grouped by prescription names. Clicking on the **Date** tab (Fig 119 **B**), the prescriptions' orders will be displayed by prescription order date and time and their status (**In progress** orders first, then **Ready**, then **Future**, then **Expired**).

The prescription orders window contains the following information:

- Full prescription strings (Fig 119 C);
- Scheduled date-time (Fig 119 D) for each order;
- Status (Future, Expired or Ready) (Fig 119 E);
- **Colored cells** (Fig 119 **F**) depending on the status of the order (gray for Future, red for Expired, green for Ready, cyan for "In progress");
- **Buttons** (Fig 119 G) to select/deselect the prescriptions' orders.

ŝ

PRESCRIPTION DUTE THEORY BUYEW		SELE	CT EXPIRED
Bugrofen – Carolve Domr. 1969 mg – Ciral – 11 (20:2013, 11:00 PM, 12:201313, 0:00 PM, 12:42013, 11:00 PM, 12:62013, 0:00 AM	19(30)2523, 11:00 PW	Febre	0
Apprin – Offernezent fable: DOSE: TBEmg – Dref – 11/26/2022 5/16 PM	11/30/2025 516 PM	60Md	01
Mattiname – Capitule 0006: 5475 mg – Gral – snok a day at 700 AH – 4ffective 10/102033, 700 AM – umit 12/31/2003, 760 AM	12/1/2013, 7:00 AM	Felare	
Glacose — mh.con DCCE: 100 mg in 1000 ml at 1 mg/h — Route of administration not applicable — Montay, Thumday, Seburday, Wednesday at 10:00 PM — until 12:06:2022, 10:00 PM	11/30/2523, 10:00 PM	Fature	°.
lee versom → Pomilier and scioent for solution for cipation 2008: 30 ml at 10 ml/h → Subsciarence → 21mms in 24m → unit 12/1/2025, 8/00 4/4	11/00/2020 6:00 PW	Ready	0
	12/1/2023, 8:00 AM	Pation	0

Fig 120

To delete expired orders:

Manually select the expired orders one by one or click on Select Expired blue button (Fig 119 H) provided to select all the expired orders at a time;

All the expired orders are automatically selected and marked with a **bin** icon (Fig 120 **A**).

Click the Update button (Fig 119 L). The Therapy Review window is automatically closed.

PREBORIPTION DATE THORNPLOY	THERE'S REVEN		
Reprofer — Capitale 1991: 399 mg — Chai — 11/30/2023, 11:00 PM, 12/2/2023, 0:00 PM, 12/4/2023, 11:00 PM, 12/4/2023, 0:00 AM	1996/2023, 11:00 PM		0.
Rapiels – Differences tablet DDBs: 100 mg – Disk – (1/30/2032, 5:18 FW	Flippipups 3/16 PM	Explored	08
Methodow – Cassule Dolla, 0.676 mg – Crui – snor a stay at 750 AM – effective, 12/1/2023, 750 AM – unel 12/91/2023, 730 AM	12/1/2012.7100.498	future	0
Oxcose — officien 1001: 100 mg in 1000 ml at 1 mg/h — fixue of administration not applicable — Monizy Thursder, Safurdey, Wednessay at 1000 PM — until 12/h2013, 1500 PM	16/30/3623, 10:00 PM	Galer	0
Bee wetern — Fonder and solvent for solution for injectory 1008: 20 eri al 10 et/h — Subsylaniaus — 2 tyrea in 24 h — unit 12/1/2023, 600 AM	(1/30/2023, 6:00 PM	Ready	0
	12/1/2022. ±00 AV		01



It is also possible to delete not-expired orders (Fig 121 A):

- Manually select the orders to be removed;
- > Click the **Update** button. The **Therapy Review** window is automatically closed.

Orders "In progress" cannot be removed (i.e. they are not selectable).

The orders selected in the **Therapy Review** window for removal won't be present anymore in the patient's prescriptions list.

A **Cancel** button (Fig 119 I) is provided to close the window without performing any update.

4.19. Closed Therapy

If the Therapy is closed, a "lock" icon - 💼 - is displayed on top of the Prescription module. See Fig 122 **A** for an example.

O'mu of classes				21
Andreasy Viscontrolleringer and a fibring to	CONSTRUES - 100 April - Incontrast Contrasts CONSTRUES - 100 April - Incontrasts Contrasts CONSTRUES - Incontrasts - DODDEDUES DOLTE - Enverse a serie (2001-0000, 0000-1100, 1000, 1100, 0000	0 #	11	ř
	NAMES (SAR) (MEL = 3 m) (m) TATALY (SAR my) in TBL and at 191,11777 mith (BBD my) of 1920/000236 = 8010234 (9),27	8.0		ľ
	- MARKAN ALMONTATION BOF TO K - MAJORITANY TORNE S & TOMANTATION A - MAINING TOM	0.2510	0.0000.0146	1
Config. 2010/ and at 25:00 Aug to 1	ANTARA TAYA NA MAMATANA - MULTING YA ANTAR YA ANTAR YA ANTAR A MANARA KA TUTI. YA MI - ANTARA MANARA MANARA	086 □		1
	ACCULUTION TIMES TOP (10) ACCULUTION	10		1
See al.				
August				
August 1				
Form.				
Name Deserve Arrange				
Para. Igen				

Fig 122

For closed therapies it is only possible to display the History (4.3) or the Overview (4.17). No action is allowed.



4.20. QR code display

A system option enables the possibility to generate a QR code for a treatment. If the functionality is enabled, a **Show QRcode** button is present on the prescription window (Fig 123 **A**).

		an anyon n 2 dayy			
(1.3-DIMENG/PROPAN-3-K_SVEROLIZINA 0.1 mgmt SOTE 2 mg in 10 ml at 2 mbh (6.8-mgNl) 5e anne			0	10401210	
3-OKCLONDEROPENENTIN PATCH - LIQUIDO USO ESTORNO IOX: 1 8 - INUN NEGLONE FEMOLALE ANTILAT - EN2OSES, 1200 PM				CONFLICTION A risks age	*
A CRAAA OTTE: 12 etg — Itti			#	1000	÷
AADAGLAAPSE'S KWINPEN 100UAR SCUIZCHE NITTABLE OTE 3 mil at 1 mille SCHTOCUTAIGA 2 times a day at 100 AM - 300 PM effect	e 6/10/2025, 2:00 PM - writt further to	0.4	:00:	ACTVR Transf	
Hature 2 1056: 4 : 50 mg to 3 : 3 mi = 9			ø	11	^
LANSE			(MARE	incris	-
4+10 mg 2+5 mi 1+25 mih 2+5 mg/t 2 mg/t		4 ACD MPI Understanding Action 0.3 mg/mg 0.6-1.5 mg	er excess smach strassenston, eres 1 mg/ml 2 - 5		
		NG			
A			Peoplet In CMM	No. Sec 11. 12	11.000
HILITORY BHOW GROOME			1017 BURPEN	A REMO	ist:

Fig 123

Click the **Show QRcode** button (Fig 123 **A**) to display the following window (Fig 124).



- > Click the button indicated in Fig 124 **A** to download the QR code.
- > Click the Close button (Fig 124 **B**) to close the window.

5. The "Therapy Execution" module

"Therapy Execution" is an application aimed at assisting the nursing staff in the tasks related to the documentation of the administration of the prescribed treatments.

The administration orders are generated according to the treatment plan specified by the physician on the "Therapy Prescription" module (see section 4). The generated orders are displayed as rectangles and placed on the "Therapy Execution" module's main screen, on a "treatment schedule" table. The nursing staff can use the available tools to document the administration of the prescribed treatments.

5.1. "Therapy Execution" module selection

To select the "Therapy Execution" module

 \succ Click the corresponding button \square on the lateral bar.

5.2. Main screen

The module's main screen is displayed. See Fig 125 for an example.

DATERNAL ALXON MUNAMME HER	EP					< NOW >	n
Moo # April			10000				
11,60	12 PM	01 PM	0.2 PNR	03 PM	D4 PM	05 994	QK PW
ACQUA FI FISIO*10F 19ML - SOU POLLIME: 10 ml - INTRAMUSCOU							
PARACETAMOLO - PREPARAZIO DOSE: 1000 mg - ORALE DILUTT	NEINETTABILE						CHIMAN
ALEXALUTIDE — 0.5 mg/ml DOLE: 1000 mg in 2000 ml at 200	0 mi/h (1000 mg/h) — 4/l/2	024, 2.20 PM	jana da a				
BUPROFENE COMPRESSE						[Northerselat and	
DOSE 1000 Mg - ORALE DILITO) Administer in case of					Contractions and	and a second
FURDSEMIDE CONTRESSE DOLL: 500 mg ORALE 4/0/20	04.413.00 PM						
	04.at 3.00.PM						
COLL: 500 mg - ORALE - 4/0/20	WHICO						
0011:500 mg - 08ALE - 6/0/20	WHICO		2460 mg	1			
COLL: 500 mg - ORALE - 4/0/20	WHICO		(2000 mg				

Fig 125

5.3. Prescribed orders representation

The treatments prescribed on the "Prescription" module are displayed as rows.



The color on the left of the row provides information on the prescription status and/or type. See section 5.3.1 for the colors legenda.

On the left of the row the available information for the prescription is specified. In Fig 126 **A**, for example, the information available is: treatment name and type, dose, administration route, administration schedule).

The rectangles on the route represent the single administration orders (Fig 126 B). See section 5.4 for the description of the administration chart.

The upper half of the row - \checkmark - can be clicked to open the administration details and commands.

For the treatments that belong to a specified class, the class is indicated in the upper-right corner of the row (Fig 126 D – "Paracetamol" belongs to the "Analgesic" class. The class of a treatment is specified during configuration.

5.3.1. Colors legenda

The different colors provide information on the prescription status and/or type.

There are 6 possible colors:

Furosemide — Oral suspension DOSE: 700 mg — Oral — 12/4/2023 at 12:50 PM	Grey characterizes active prescriptions whose orders must be administered in the future.
Brilliant green — Faint DOSE 1000 mg — Cutameous — 12/4/2023, 10/19 AM	White characterizes completed prescriptions.
Paracetamol — Soluble tablet DOSE: 1000 mg — Ocial — 12/4/2023 at 11:00 AM -5:00 PM	Green characterizes prescriptions having an order "ready to be administered".
Polihexanide — tye drops DOSE: 500 mg — Ocular — 12/4/2023, 10:12 AM	Red characterizes prescriptions having at least one "late" order.
Albigrutide — Powder and solvent for solution for injection DOSE 20 ml at 10 mi/h — Subcutaneous — 12/4/2023, 10:15 AM	Cyan characterizes durative prescriptions having one order in progress (see paragraph 3.3 for a description of durative prescriptions).
Aminosalicylic add — Gastro-resistant granules DOS2: 500 mg — Ocal — Admenster in case of	Purple characterizes conditional prescriptions; these are treatments to be administered only if specific conditions occur.

The boxes are displayed "by urgency" on the "Administration" screen. The "late" orders are on top, then the "ready" ones, then the "durative" orders in progress, then the "conditional" ones, then those to be executed in the future. In the end are the boxes corresponding to completed prescriptions.

Therefore, the colors appear in this order, from the top of the screen:

- 1. red
- 2. green
- 3. cyan
- 4. purple
- 5. light grey
- 6. white

The treatments can be sorted otherwise using the buttons on the bottom-right corner (Fig 127 **E**, described in paragraph 5.11).
5.4. The orders chart

The administration screen displays on a chart all the orders already generated (the already executed ones, the future ones and the "in progress" ones - Fig 127).

11 AM	OT PM	/02. PM	03. FM	(84 PM	45 194	06 P
ACQUA PLESSO*10F 10ML - SOLVENTE USO PARENTER VOLUME 10 ml - INTRAMUSCICLARE - 4/9/004, 2:22						
Automic (emi - micemicacionis - envires, 222)						
PARACETAMOLO PREPARAZIONE INETTABLE						C ALIGNET D
DOSE 1900 mg = OBALE DELUTO = 4/6/2024, 2:27 PM						Canadrante
ALBIGLUTIDE - 0.5 mp/ml						
DO11: 1900 mg in 2000 ml at 2000 ml/h [1900 mg/h] -	40872024, 200 HM	2388 with	10			
IBUPROFENE — COMPRESSE DOSE, 1000 mg — OVACE DILLETO — Administrar in case	ol				borreivenet 24	anti-traffacentiatory
FUROSEMIDE - COMPRESSE						
DOLE: 560 mg - OBALE - 4/8/2024 # 3:00 PM			100			
ADDD FITICO/COMPLESSO VITAMINICD DOSE: 2000 mg — 4/8/2024, 2:16 PM						
		2000 mg				
		State Providence				

Fig 127

The orders are positioned in a grid, made of rows and columns. The orders corresponding to the same treatment are on the same row (in Fig 127 there are six prescribed treatments on the left - Fig 127 **A**); the columns correspond to the hours of the day.

The red bar (Fig 127 **B**) indicates the present time. The screen scrolls left as time goes by, thus the bar indicates the present moment on each row. In Fig 127 it is 2:30 PM approximately. The time can be read on top (Fig 127 **C**). The date is displayed in the top left corner (Fig 127 **D**).

5.4.1. Graphic representation of the orders

The colored rectangles represent the different orders, positioned in the place corresponding to their prescribed administration time (or execution time if already executed).

5.4.1.1. Administration tolerance period

The length of the rectangles is proportional to the period that was indicated as "tolerance period for the administration" when the order was prescribed (see section 4.15).

For instance: if the treatment is prescribed at 13:00 and the administration tolerance period is set to 30 minutes, the administration time indicated on the chart is a period going from 12:30 to 13:30.

The position of the rectangles corresponds to the prescribed administration times. The rectangles length is proportional to the tolerance period.

5.4.1.2. Order status

The rectangle color indicates the status of the order.

OSE: 500 mg — Oral — Administer in case of	
aracetamol — Soluble tablet	
OSE 1000 mg - Oral - 12/4/2023 at 11:00 AM - 5:00 PM	
arosemide — Oral suspension	
OSE: 700 mg - Oral - 12/4/2023 at 12:00 PM	
rilliant green — Paint OSE: 1000 mg — Cutaneous — 12/4/2023, 10:19 AM	
the last of the second se	

Green means that the order is to be administered immediately, i.e.: it means that it is within the administration tolerance period. When the rectangle is green the vertical time bar intersects the rectangle (Fig 128 **A**).

Light grey means that the order is to be executed in the future. When the rectangle is light grey the vertical time bar is on the left of the rectangle (Fig 128 **B**).

When a treatment is administered, a vertical grey bar is positioned on the prescription row, in the place corresponding to the exact administration time. The administered dose is specified nearby. The vertical time bar is on the right (Fig 128 C).



The length of the boxes corresponding to an executed order is reduced to indicate the exact administration time. The tolerance period is not displayed anymore. When the prescription is durative (a drip, for instance) the administration duration is fully displayed in grey. See paragraph 5.5.2 for the durative orders administration procedure.

Red means that the administration of the order is late, i.e.: the tolerance period is over (Fig 129 **A**). When the rectangle is red the vertical time bar is on the right of the rectangle.

Cyan characterizes durative administrations in progress (Fig 129 **B**). The vertical time bar in these cases coincides with the rectangle right side. The administered dose is specified inside the rectangle, on the left. A "Stop" order can be generated for durative administration if the administration duration is specified at prescription time. The "Stop" order is positioned according to the scheduled duration of the administration (Fig 129 **C**).

Polihekanide – Eye drops DOSE 500 mg – Ocular – 12/4/2923, 10:12 AM		
Albightide — Powder and solvent for solution for injection DOSE 20 ml at 10 ml/h — Subcutaneous — 12/4/2023, 10:15 AM	B	O
Aminosalicylic acid — Gastro-resistant granules DOSE: 500 mg — Oral — Administer in case of		
	-ia 129	

The duration of an administration is displayed in cyan. When the administration is stopped the duration becomes grey (Fig 130). Possible changes in the administration dose are specified inside the rectangle.



When the icon is displayed on the left of a rectangle (Fig 131), it means that the order has not been validated yet or is expired. See section 3.3.3 for the explanation of the "order validation" concept. The non-validated orders can be administered using a specific procedure. This procedure is described in section 5.5.7.





"Late" orders (the red ones) expire after a time that is defined by the ExpiredOrderTimeoutHours system option (see the document DSO ENG System Options for more information).

5.5. How to record the treatment administration

5.5.1. Punctual treatment administration

To record the administration of a treatment,

 Click the upper half of the row corresponding to the treatment to be administered (Fig 132 A).

		(A)	
Paracetamel — Soluble tablet DOSE: 1000 mg — Oral — 2 times a day at 11:00 AM - 5:00 PM — until fuctive notice		v	~
	Fig 132		

The row enlarges to show the prescription summary and the administration commands (Fig 133).

Paracesamol — Solutile tablet. DOSE: 1000 mg — Deal — 2 times a day at 11:00 AM - 5:00 PM — um	Turther notice	
	SHOP TAMAY EXECUTE	
	SUMMARY	
	toco mg 57 kg	
		Prescribed by ADMN on Too, Dec 05, 10:15 AM
	F: 400	

Fig 133

Click the Execute button (Fig 133 A).



The "summary" indicated in Fig 133 **B** refers to the prescription values, not to the values of a specific administration. Therefore, if the values of an administration change, the summary values remain the same.

The following window is displayed, making it possible to confirm or change the administration values. The window is described in section 5.8.

			TODAY XT 11:00 AM	10):18 A	MA
100	0		DUMMONIY		12/5/2023	
mg			1000 mg 57 kg			
				15	35%/4	their
4	12	3	area (•	•
*	18	- t.				
a)	+	1.0				
ŭ	iii.					
8	a					0

> Click the **Administer** button to record the administration of the treatment (Fig 134 A).

The rectangle corresponding to the specific order changes in the following way (Fig 135 A – a vertical grey bar remains in the position corresponding to the administration time; a label indicates the administered amount).

Paracetamol — Soluble tablet	at 11:00 AM - 5:00 PM — until further notice	Ø
see to still star a since any t	STORE THE SHORE THE SHORE THE DECEMBER	1000 mg
	Fig 135	

The treatment administration is this way recorded.



The order does not need to be in "Ready" status (green color) to be administered. The administration can be recorded before (or after) the prescribed administration time (light grey or red). Specific user permissions are required to perform these tasks. The reason for anticipating or delaying the administration must be explained in a note (see section 5.5.7).

5.5.2. Durative treatment administration

For durative treatments, it is necessary to record the beginning and the end of the administration.

To record the administration of a durative treatment:

Click the upper half of the row corresponding to the treatment to be administered (Fig 136 **A**).

					A	
	— Infusion	Parity of administration	and the second states of	12/2/2022 11:12	• • • •	
DUSE: 10 I	ng in 500 ml at 10 mg/h –	- Koute of administration	on not applicable -	- 12///2023, 11:1/	C-BM	
			Fia	136		

The row enlarges to show the prescription summary and the administration commands (Fig 137).

Morphine — Infusion DOM: 10 mg in 500 mil at 10 mg/h — Rouse of admit	instation not applicable	- 12/7/292	1.11.17 AM					^
					A			
			50P 12/7	10044 /2023, 11:17 AM	EXECUT	ΰ.		
				SUMMURY				
	attensor 10 mg	untiave 500 ml	coveninares 0.02 mg/ml	sens 500 mi/h	sea ang 10 mg/h	the state	would 90 kg	
								Prescribed by ADMIN on This Dec GE 11:36 AM



Click the Execute button (Fig 137 A).

The following window is displayed, making it possible to confirm or change the administration values. The window is described in paragraph 5.8.

500	8			TODAY AT 11 SUMMA				:19 A	
mi/ħ	a.		10 mg	500 mi	0.02 mg				•
2	2		into 500 mirts	to organ	maninipe 171	90 Ng	•	30000	-
		G	×	A0103		2			
Эř.	24	e							
a.	-Ø								
									6
								WHICEL	

> Click the Start button to record the beginning of the administration (Fig 138 A).

The rectangle corresponding to the specific order becomes cyan. A cyan bar is drawn on the chart while the administration goes on; a label indicates the infusion rate (Fig 139 **A**).

If, at prescription time, a duration is indicated for the administration, a stop indication is created on the row according to the indicated values. The "Stop" rectangle, if present, is positioned according to the infusion total duration (Fig 139 **B**). If the exact duration of the administration is not indicated, then the "Stop" rectangle is not present. The administration procedure remains the same.

			0			C) CHANGE	SEDP		6	
			V						V	
			SUMBARY						INVERTING	
ANDAR	VOLUME.	disconness	1000	100020920	Diakyvelie	(CONT)		AMOUNT	121044	DALAATYON
10 mg	500 ml	0.02 mg/ml	500 mi/h	10 mg/h	1m	90 kg		-0.091 mg	-4.551 ml	-1m

Fig 139

The prescribed administration values are displayed on the left (Fig 139 C - Summary). The administered values are displayed on the right (Fig 139 D – Amount, Volume, Duration – these values are updated in real time).



The "summary" indicated in Fig 139 C refers to the prescription values, not to the values of a specific administration. Therefore, if the values of an administration change, the summary values remain the same.

If the "Stop" rectangle is present, when the red "now" bar intersects it, it turns green, meaning that user action is required (Fig 140 **A**). From this moment on, the "Change" button

is not available anymore. Only the Stop button is available (Fig 140 B). The tolerance period for stopping the infusion "in time" is 15 minutes (configurable).

If the "Stop" rectangle is not present, the "Change" option remains present.

Marphine — Infusio DOSI: 10 mg in 500		h — Route of admi	nistration not a	ppficable — 13/	7/2023, 11 17	AM and	Y				×
						₿	\$70H				
			SUMMARY						AUM/NING		
Meptr 10 mg	ecune 500 ml	Decomentes 0.02 mg/ml	urus 500 mi/h	10 mg/h	toistos 1h	enori 90 kg		esculet -10.107 mg	V0.048 -505.336 ml	pulaition -1h & 1m	
										esothar by ADMW ini Thu, i	0xc32.77.1634M
						Fig	140				

In both cases:

Click the STOP button to stop the durative administration.

An administration window is displayed (Fig 141). If required, it is here possible to insert notes or operate on the recorded administration time (see paragraph 5.8).

Annie Information in annie i	100AY A7 2	00.994		2	:10 P	M
II Strain Hav	140910		123		12/7/2021	
is 20min Nov						
				IN	30min	104

Click the **Stop** button on the administration window (Fig 141 A).

The administered order remains as a dark grey rectangle on the chart, its length corresponding to the actual duration (Fig 142 A).

Morphine — Infusion DOSE: 10 mg in 500 ml at 10 m	g/h — Route of administrat	ion not applicable — 12	500 mi/h	
		=: 440		

5.5.3. Drugs interactions

If there are interactions between the current administration and other prescribed treatments, a warning is displayed below the summary, indicating the involved treatments and the interaction type. See, for example, Fig 143 **A**. Drug interactions are defined during configuration.

		907	1000 4/10/2024,	D-	EXECUTE		
			SUMM	IARY			
	securi 500 mg	emine 1000 ml	containercos 0.5 mg/ml	sisan 200 mi/h	онин элен 100 гед/1)	75 Kg	
NUTRE - THE POLINE BOARD TO	<u> </u>						

Fig 143

5.5.4. Allergies

If allergies are indicated for the patient, a flag is present on top of the Execution page (Fig 144 **A**). If a treatment which the patient is allergic to is administered, the "Allergy" is notified before the administration (Fig 144 **B**). It is here possible, not mandatory, to add a note. Patient allergies are indicated outside of the Therapy Web application, on other modules.

BORISTAT BI	LL, PATIENT & C" THE M VIEW IN A	n juq nu mari		ß	A -11411 AL -114
		INUFROFINE - CONSERVE DOI: 0.000 0.000 0.000 - 0.000000.006 //	8.	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	e 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
PE Mirech	11 THE THY CAR LAN AN THE - COMPANY FS - ONLY DURING - REPORTS AND TH	plines in	WARNING and a reduce research the units of an the objective	17.0	,
0 41 42 43 43 43 44 44 44 44 44 44 44 44 44 44			Long same of high	8	
B 4010.0	17044 - C47542 Mg - M7644-X25648 - 220004 200744				
	ападарын та — 40.004 (0.00 мм			CANCEL	(1999) -
100.410	eg – ann 1111 a 200 400 400 fer - 5700000	x 10.00			
THERAPY					0.411 LATERING BOUT 114.0

Fig 144

5.5.5. End a durative administration before or after the scheduled time

If there is not a "Stop" rectangle, the system does not provide any indication on the administration stop. If there is a "Stop" rectangle, there are cases in which the administration is stopped before or after the scheduled time.

If the administration is stopped before the scheduled time (i.e. when the "Stop" rectangle is grey, positioned on the right of the red "now" bar – see for an instance Fig 141 **A** and **B**), then, after clicking on **STOP**, the user is required to specify in a note the reason for anticipating the end of the administration. The note is then displayed in the administration window (Fig 141).

The tolerance period for stopping the durative administration "in time" is 15 minutes (configurable). After 15 minutes, the "Stop order" is late and the "Stop rectangle" turns red (Fig 145 **A**). The administration goes on.

		100 001					_				
						-	ITOP .				
		1	SUMNOARY						RUNNING		
autourt 100 mg	viculati 1000 mil	commission 6.1 mg/mi	inus 500 ml/h	teog sesan 50 mg/h	2004/104 271	would 57 kg		-137.775 mg	eiatani. -1377.751 ml	ouwron 2h & 45m	

Fig 145

To stop an administration after the scheduled time, also, the user is required to specify in a note the reason for delaying the end of the administration. The note is then displayed in the administration window (Fig 141).

In both cases, the administered orders remain as a dark grey rectangle on the chart, their length corresponding to their actual duration.

5.5.6. Changing the administration values

It is possible to change the administration values of a durative administration while the administration is in progress. To do that:

 \succ Click the upper half of the row (Fig 146 **A**).

The administration details and commands are displayed (Fig 146 B).

						•				
						\$ CHWNEE	Si06			
			SUMMARY						RUNNING	
waxui 10 mg	source 500 mil	0,02 mg/ml	seusi 500 mil/h	10 mg/h	materiae Th	uttion 57 kg		-2.677 mg	vittine ~133,866 ml	-1.6m

Fig 146

Click the CHANGE button (Fig 146 C).

The administration details window is displayed (Fig 147).

500)		NOTE	12	9	:24 A	
mi/ħ						•	
3	2	880			18	30min	10m
4	5	6					
17	0.	1990					
. 11	¢						
u	c						
4	c				CA	NCEL	CHANG

Use either the virtual keyboard or the workstation keyboard to insert the new value in the field indicated in Fig 147 A.

The window changes as shown in Fig 148. The user is required to indicate in a note the reason for changing the administration values (Fig 148 **A**).

700			NUTES	9	:24 A	
ml/h			ance changing value	• 16	A Blance	1000
्रम	3	2. 3 0			•	-
4	4	(e)				
3	8) (#])				
0	c					
						6

Fig 148

> Click the **CHANGE** button (Fig 148 **B**).

The administration chart changes as follows. The new speed is indicated in a label (Fig 149 **A**).

							No. of Concession, Name			
						C CHANNEE	вота 🔳			
			SUMMARY						RUNNING.	
uicoit 10 mg	soline 500 ml	towgeneoos 0.02 mg/ml	sees 500 mi/h	onuconno 10 mg/h	townoi: Th	econ 57 kg		медот ~3.26 mg	voluer ~163,012 ml	ouende >19m
									0.	escribed by ADMIN on Thu, Dec 07.



The change is signaled on the "Prescription" module by the icon $\hat{\circ}$ on the prescription table.

5.5.7. Prescribed changes of a running administration

The necessity to either change the values of a running durative administration or to stop it can be prescribed on the Prescription module (see paragraph 4.8). In these cases, a "Change" or "Stop" order is created on Execution, on the administration row.

"Change" order

See, for example, Fig 150.

FUROSEMIDE — 0.5 mg/ml — SOLUZIONE PER INFUSIONE DOSE: 500 mg in 1000 ml at 200 ml/h [100 mg/h] — ENDOVENOSA DILUITO — 4/10/2024, 11:28 AM	A	
200 mil/h		
	250 mith	
Fig 150		

In Fig 150 the administration of Furosemide is running with a speed of 250 mL/h. On Prescription, the physician prescribes to change the speed to 300 mL/h. A green rectangle, with 15 minutes tolerance, is displayed on Execution, informing the nursing staff that an action is required for that treatment. The treatment status turns to "ready" again (green). See Fig 151 **A**.

FUROSEMIDE — 0.5 mg/ml — SOLUZIONE PER INFUSIONE DOSE: 500 mg in 1000 ml at 200 ml/h [100 mg/h] — ENDOVENOSA DILUITO — 4/10/2024, 11:28 AM
200 mi/h Z50 mi/h
Fig 151

To document the administration change:

Click the upper half of the row (Fig 152 A).

The administration details and commands are displayed (Fig 152 B).

						ice :-	FODM? 4/10/2024, 2:58, FM	UPDATE				
>			TUNM	ANY						RANNING		
	same and	022.0448	CONTRACTOR .	9970	10410.0101	weinet			AMOUNT	ART LINE	DAMAGED	
	500 mg	1000 mil	0.5 mg/mi	200 mi/h	100 mg/h	75 Kg			-294.6 mg	~589,1 ml	~2h & 39m	

Fig 152

Click the **UPDATE** button (Fig 152 **C**).

The administration details window is displayed (Fig 164).



The new administration values are read-only (Fig 164 **A**). It is possible to adjust the execution time and to add a note, if necessary (Fig 164 $\mathbf{B} - \mathbf{C}$).

Click the **CHANGE** button (Fig 164 **D**).

The administration chart changes as follows. The new speed is indicated in a label (Fig 165 **A**).

FUROSEMIDE - 0.5 mg/ml - SOLUZIONE PER INFUSIONE DOSE: 500 mg in 1000 ml at 200 ml/h [100 mg/h] - ENDOVENO	SA DILUITO — 4/10/2024, 11:28 AM	A
200 mi/h	250 mi/h	300 mi/h
	Fig 154	

If the prescribed change is not performed in time (i.e. within the tolerance period), the green "change rectangle" turns red, meaning that there is a prescribed action that is late (Fig 155 **A**). The treatment status turns to "late" (red); the infusion goes on with the current values. Then, when the change is made, it is mandatory to specify in a note the reasons for not performing the action when prescribed.

FUROSEMIDE — 0.5 mg/ml — SOLUZIONE PER INFUSIONE DOSE: 500 mg in 1000 ml at 100 ml/h [50 mg/h] — ENDOVENOSA DILUITO — 4/11/2024, 1:04 PM		0		
	100 mi//h	M		
Fig 155				

If the prescribed change in the administration value is not performed, the user can document the fact that the action was skipped.

To do that:

> Click **SKIP** instead of **UPDATE** on the administration chart (Fig 152 **D**).

It is then mandatory to specify in a note the reasons for not performing the prescribed action (Fig 156 \bf{A}).

	WARNING	
please in	nert a role with the reason you are canoding this und	ir -
Note for skipping. (meets solution	团
		CANCEL

Click **NEXT** (Fig 156 **B**).

Further confirmation is required. The window changes in the following way (Fig 157).

	TODAY 2:35 PM	
	NUTE	12
	Search Lance Hore	
Note for skipping.		
		•
		(A)

Click **DO NOT UPDATE** (Fig 157 **A**).

The prescribed update is not performed, the reason for not updating is recorded. The administration chart returns to the original state (Fig 150).

"Stop" order

See, for example, Fig 158.

FUROSEMIDE — 1 mg/ml — SOLUZIO DOSE: 1000 mg in 1000 ml at 1000 ml	<pre>ie PER INFUSIONE h [1000 mg/h] — ENDOVENOSA DILUIT</pre>	IO — 4/17/2024, 1:31 PM	
		1000 mi/h	
	Fig	158	

In Fig 158 the administration of Furosemide is running with a speed of 1000 mL/h. On Prescription, the physician prescribes to stop the administration. A green rectangle is displayed on Execution, informing the nursing staff that an action is required for that treatment. The treatment status turns to "ready" again (green). See Fig 159.

FUROSEMIDE — 1 mg/ml — SOLUZIONE PER INFUSIONE DOSE: 1000 mg in 1000 ml at 1000 ml/h [1000 mg/h] — ENDOVENOSA DILUITO — 4/17/2	1024, 1:31 PM
Fig 159	

To document the administration stop:

 \succ Click the upper half of the row (Fig 160 **A**).

The administration details and commands are displayed (Fig 160 **B**). Only the "Stop" button is available (Fig 160 **C**).

-	ove mg m ress	ini at 1640 mil	h [1000 mg/h] — E	Industria and	alla – arranga				V	
						C	PF.			
>			SUM	NAR					BUNNING	
	succent 1000 mg	voulie 1000 mi	iniceseinen 1 mg/ml	ires 1000 mil/h	oscerosent 1000 mg/h	uuumou 1h		aattosit ~341,6 mg	===== >∃41.6 mi	summer -20m
				сня Ка						
-	ment saided in	a for the day								Prescribed by ADMIN Wed, Apr 17, 01:31 m

Fig 160

Click the STOP button (Fig 160 C).

The administration details window is displayed (Fig 161).

	TODAY 151 PM	A	> 1	:54 P	
>	22	8			
			- 10	Itees	1

It is possible to adjust the stop time and to add a note, if necessary (Fig 161 $\mathbf{A} - \mathbf{B}$).

Click the STOP button (Fig 161 D).

The administration chart changes as follows. The length of the grey rectangle corresponds to the actual duration of the administration (Fig 162 C).

FUROSEMIDE — 1 mg/ml — SOLUZIONE PER INFUSIONE DOSE: 1000 mg in 1000 ml at 1000 ml/h [1000 mg/h] — ENDOVENOSA DILUITO — 4/17/2024, 1:31 PM
1000 milite
Fig 162

5.5.8. Administration of an order either expired or not-yet-validated

The order validity expires after a certain time (see paragraph 3.3.4 for more information). An expired order can be administered anyway but requires specific user permissions and a specific procedure. Expired orders are characterized by the Ψ icon (Fig 163).

1	Paracetamol — Soluble tablet DOSE: 1000 mg — Oral — 2 times a day at 11:00 AM - 5:00 PM — until further notice
l	
	Fig 163

To administer an expired order (or not yet valdated):

Click the upper half of the row. The administration details and commands are displayed (Fig 164).

Paracetamol — Solubie tablet DOGE 1000 mg — Oral — 3 times a day at 11:00 AM - 5:00 PM	-unil further nation	
	4049.400 (377.0033.500 PM EXECUTE	
	SUMMARY Michine Benner 1000 mg 57 kg	
		Prescribed by ADMIW on Tue, Dec 05, 10:15 AM

Fig 164

Click the Execute button (Fig 164 A).

A window opens, requiring the user to insert the reason why an expired order is administered.

- Specify, in a note, the reason for administering an expired order.
- Click Next.

The administration detail window is then displayed. The note is displayed on the window (Fig 165 \bf{A}).



Click **Administer** to administer the order.



The same procedure can be used to record the administration of a future not-yetvalidated order.

5.5.9. Skip administration

To document the fact that a certain treatment was not administered:

Click the upper half of the row. The administration details and commands are displayed (Fig 166).

Polihexanide — Eye drops DOSE 1000 mg — Doular — 12/13/2023, 10:13-AM		~
	SIGP TOONY EXECUTE	
	SUMMARY	
	ticco mg 57 kg.	
		Prescribed by ADWW on illest, Dec 13, 70:13 AM



- Click the Skip button (Fig 166 A).
- Specify, in a note, the reason for not administering the treatment (Fig 167 A).

	WARNING		
	please insert a note with the reason you	and concepting this under	
Fields for skipping the ad	servic rail rich		9
			(

Click Next (Fig 167 B).

The administration detail window is displayed. The note is displayed on the window (Fig 168 **A**).

	TODAY AT 10:13 AM	
	, laitte	100
Row to skyping the administration of	a freetneret	8

> Click **Do Not Administer** to skip the administration (Fig 168).

5.5.10. External resources (Help)

If the "Info" icon is present on the top-right corner of the "Summary" area of the treatment administration row (Fig 169 **A**), a link is available to configured external resources (an external URL or a file - the drug leaflet, for example).).

Contraction (0
	Ō
	Annal and a second s

Fig 169

Click the icon to access the external resource.

The resource accessed is the same triggered by the same icon on the Therapy Prescription module (see paragraph 4.14).

5.6. "Range" administration

If the treatment was prescribed using the "Range" option (see paragraph 4.15.2), the prescribed ranges are indicated on the administration row.

Manuel 2 Martine 1 - Startine 1 - Startine -	9	
A titreng 2-5 ref 1-25 might 2-5 mig		R
	All regime Adv 13 reg	Freedoal by attent for part 11 11



The global ranges are indicated in Fig 170 **A**. In case of mixtures, the ranges of every component are displayed on the right (that is the case indicated in Fig 170 **B**).

Click Execute to administer the treatment (Fig 170 C).

The administration detail window opens (Fig 171). The range is indicated in Fig 171 A.

A COMPT AND ADDRESSINGS	•	
1 3 3 4.400 MPT 44 ADMB BRACK • 1 3 3 0.3 mg/mg U-6 mg 1 mg/ml 2 mg • 2 4 4 • <		
1 3 3 0.3 mg/mg 54 mg 10 mg/mg 10 mg/mg 4 5 4 2 4 4	•	-
• •		
· · ·		
· · ·		
a c .		

Fig 171

Use the administration procedure as described in the previous paragraphs to administer the treatment. If the administration values are changed within the prescribed range, the "Value changed" note is not required (if it was, according to configuration).

5.7. Administration with double signature

A treatment can be configured to require a double signature at prescription time and/or at administration time. For administrations, it is necessary that a second user, different from the one that is administering the treatment, validates the administration. The administration procedure is the same described in paragraph 5.5 but, if a double signature is required for

the administration, a specific icon - - is displayed on the rectangle representing the order on the Execution plan (Fig 172 A).

		I	Fig 172	
	Before signature, the i cons (Fig 173).	rectangle rep	resenting the order of	can be characterized by tw
U		08:00		12:
			Fig 173	
	Гhe ♥ icon indicates t Гhe ❷ icon indicates t			

For the signature procedure at prescription time see paragraph 4.16.

Two different procedures can be configured to allow a second user to sign the administration, depending on the *ExecutionSignMode* system option.

If *ExecutionSignMode* = 0 the signature must be performed on another Digistat session. That means that the administration must be signed by another user logged on a different Digistat session.

If *ExecutionSignMode* = 1 the signature must be performed within the same Digistat session, where the credentials of a different user are requested.

Both cases are described below.

5.7.1. Signature on a different Digistat session

In Fig 174 the Abacavir administration is indicated. In this configuration, Abacavir requires a double signature (Fig 174 **A**).

Abacavir — Tablet	
DOSE: 10 mg - Oral - 09/04/2025, 12:36	
	Fig 174

If the *ExecutionSignMode* system option is set to 0, use the usual administration procedure, described in paragraph 5.5.

The double-signature mandatory requirement is indicated on top of the administration window (Fig 175 **A**).

10			This prescription requires a double sign	0		12:50	
			0661 - 1236				
ng	2	3	waryan wester 10 mg 80 kg		Th	10min	1mk
4	5	0	note	(3)			100
7	8	9					
0	c						
							B
						C.c.c.	V

Click the Administer button (Fig 175 B).

The following window opens, warning that a second signature is required (Fig 176).



To sign the administration it is necessary to log in with different credentials on a different Digistat session and select the same patient.

On the row corresponding to the administration to be signed, the **SIGN** button will be displayed for the second user. See Fig 177 A.

On the bottom-right corner, are displayed the details of both the prescription and the related signature (if a signature is required at prescription time, if not, only the prescription details are displayed - Fig 177 **B**).

Abscave — Tablet DDBE: 50 mg — Oral — (9)/04/2025, 13:29	•	<u>e</u> ~
	08/04/2025, 13.29	
	50/6001	
	aantoole entroor 10 mg 80 kg	
		B Synchology 400 - mar 08 apr, 12,25 Synchology - mar 09 apr, 13,26
ниятски		

Fig 177

Click the **SIGN** button (Fig 177 **A**).

An administration summary window is displayed (Fig 178).

	OGGI - 13:30		
	would would		
	10 mg 80 kg		
	NOTES		
ini c	otes are present inside f	tvis order	

Click the **SIGN** button (Fig 178 **A**).

The order will be this way signed and displayed as administered on the Therapy Execution plan (Fig 179 **A**).

Abacavir — Tablet DOSE: 10 mg — Oral — 09/04/2025, 13:29	A 10 mg
Fig 179	

5.7.2. Signature on the same Digistat session

In Fig 180 an Abacavir order is indicated. In this configuration, the administration of Abacavir requires a double signature (Fig 180 **A**).

Abacavir — Tablet DOSE: 10 mg — Oral — 09/04/2025, 12:36	
	Fig 180

If the *ExecutionSignMode* system option is set to 1, use the usual administration procedure, described in paragraph 5.5.

The double-signature mandatory requirement is indicated on top of the administration window (Fig 181 **A**).

10			This prescription requires a double sign	0		12:50	
ng			0661 — 12:36				
4	2	з	warsani wester 10 mg 80 kg		1h •	10min	lmir •
4	5	0	90156	æ			
7	8	9					
0	с	10					
							െ
							D

Click the Administer button (Fig 181 B).

The following window opens, containing a "Sign" area that makes it possible to double-sign the administration (Fig 182 **A**).

	OGGI - 13:48	
	WATHOUT WHILE-W	
	10 mg 90 kg	
	NOTES	
	no notes-are present inside this order	
-		
-	SIGN	l a
Usemame	SIGN	
Usemame		

Insert the credential of a second user (different from the one currently logged) in the Username and Password fields (Fig 183 **A**). The **SIGN** button will be enabled (Fig 183 **B**).

	OGGI - 13048	
	Addressed weight	
	10 mg 90 kg	
	NOTES	
	no notes are present inside this order	
	no notes are present inside this order.	
	ing notes are present inside this order.	
	ing notes are present inside this order.	
	na notes are present inside this order. SIGN	
	SIGN	

Click the **SIGN** button (Fig 183 **B**).

A confirmation window is displayed (Fig 184).

11.00	
UPDA	TED
Your order has been	succesfully signed
	A
~	
	-
PRESCRIPTION IN	FORMATIONS
Drug nume	Abacavir Tablet
Scheduled order time	13:48
Scheduled order time	1,2,40
Fig 1	

Click the CLOSE button to close the window (Fig 184 A).

The order will be displayed as administered on the Therapy Execution plan (Fig 185 A).

Abacavir — Tablet DOSE: 10 mg — Oral — 09/04/2025, 13:48	1 a mg
	Fig 185

5.8. The administration window

The treatment administration window (Fig 186) allows to edit some of the values of the administration on the "Therapy Execution" module.



The prescribed values are indicated on top (Fig 186 A).

In the central area of the window the following information is displayed:

- Administration scheduled time (Fig 186 B).
- Administration summary (Fig 186 C Note: the values specified in this area depend on the treatment type).
- Generic user notes (Fig 186 D).
- Notes related to: late/early administrations; changes in the administration values; changes in the administration time (Fig 186 **E**).

On the left, a numeric keyboard allows to change the administered quantities (Fig 186 F).

Which parameter is displayed on the treatment administration window depends on the configuration of the treatment (for infusions, for example, it could be volume, speed, drug speed etc.). If "Speed" is chosen as main parameter, then the "Drug speed" value is also displayed below. If "Drug speed" is chosen as main parameter, then the "then the "Speed" value is also displayed below (Fig 187 **A**).



To change the administration amounts:

- Use the virtual keyboard to specify the new amount. The window changes as shown in Fig 188.
- > Type a note explaining the reason for the changes, if required (Fig 188 B).

The treatment administration mode can be configured to require a mandatory note, an optional note or to not allow the amount change.

Click the Administer button (Fig 188 C).

Arrest			YESTERDAY AT 11:05 AM		12	2:55 F	M
800			ECINIMADA.	6		12/12/202	5
mg			800 mg 57 kg				
					29	20mm	tów
<u>_1</u>	÷.	- 90	HOTO		*	•	
4	8	ε.					
		. (actint transmit	8			
-	-	-	LATE ADARBETTELETING	œ			
0	6	<u>. 8</u> ,	Late advectation role				
							ര
_							V

The time display on the right (Fig 188 **D**) should indicate the time at which the treatment is actually administered. The time displayed by default is the time at which the administration window opens. If the administration of an order is documented at a time that is different from the time of the actual administration, it is possible to change the displayed time and indicate the correct administration time.

To do that:

➤ Use the arrow buttons placed below the time display (Fig 188 D).

The upward arrows add 1 hour (left arrow), 30 minute (center), 10 minutes (right) to the time displayed.

The downward arrows subtract 1 hour (left arrow), 30 minute (center), 10 minutes (right) to the time displayed.

The window changes in the following way (Fig 189).

800			VESTERDAY AT 11:00 AM	1:05 PM						
mg			BEO mg 57 kg							
	1.00				36	30min	Tiletin			
30	2	8.:	NETHY	123	-	•				
4	5	0								
	_		SALE DAVAGED							
7	.0	0.1	Reason for changes							
0	c	B	Так нимен Тем сладал нак	10						
			Law edworptation now	12						
						(0			
иск					CANCE	1000				

> Type a note, if required, explaining the reason for the time changes (Fig 189 B).

Ø

The treatment administration mode can be configured to require a mandatory note, an optional note or to not allow the time change.

Click the Administer button (Fig 189 C).

The administered order bar or rectangle (in case of durative administration) is placed in the chart according to the time and values specified on the administration window (Fig 190).

1000 mg	500 ml/h
Fig 190	

A configuration option allows to display, for infusions, a "Delivered by a connected pump" flag. If the flag is checked, it means that the treatment is administered by a pump that is already connected to the "Digistat Fluid Balance Web" application and, therefore, the administered quantities will not be exported for the fluid balance calculation. If not checked, the pump is not connected to the "Digistat Fluid Balance Web" application and the administered quantities will be exported for the fluid balance Calculation (Fig 191 **A**).

2 mi/h		2 mg 5	nil 0.4 mg/ml	2 milth	19.2 mg/d	12	2:24 P		
0+ing special	£						10	Sterier	1mi
15.2 mg/d	1			80751		000	•		
1	2	- 81							
4	. 85	4							
1	+								
0	e.								

Fig 191

5.9. "Execution" module prompts

On the upper left corner of the "Execution" screen different icons can be displayed as prompts for the user relating to the overall status of the treatment plan (Fig 192 **A**).

Min 4 December			l'						
Men 4 December	OB AM	09.446	10 AM	TT MM	12.PM	ST P	i.	1	02.P1
Politiesanide - Epitemian DONE: 500 mg - Druker - 12/4/2023	10:12 AM								
Altiglutide — Powder and solverti for 005E-20 mE at 10 mil/h — Sobcutane	solution for injection								
COSE_20 meat 10 mem 20000000	505 - 1209/2023, 30/15 AM		Henth						
Antinosalicylic acid — Gantro-resistur DOSE 500 mg.— Oral — Askninister in	t granules s case of								
Paracetamol — Sciuble tablet 005E: 1000 mg — Oral — 12/4/2023 a	n 11/00 AM - 5:00 PM			_					
Furosemide — Oral suspension DOSE 700 mg — Oral — 12/4/2023 at	12:00 PM			_	_				
Brilliant green - Paint POSE: 1000 mg - Cutamenus - 1244	2023, NO 19 AM								
			100 mg						

Fig 192

In Fig 192 **A**, for example, the icons indicate that on the treatment plan there are:

- 1 order expired (red)
- 1 order ready (green)
- 1 order running (cyan)
- 1 order pending (grey)

See, in Fig 193, an enlarged view.



5.10. "Execution" time range display

The commands available in the top-right corner of the "Execution" screen allow to change the time range displayed (Fig 194 \bf{A}).

Wen 4 December			100000			CONTROL 1	
-07.004	OB AM	03.446	10 AM	11 MM	12.PM	ST PM	-02 PM
Polihesanide — Eye drugs. DOM: 500 mg — Druke — 1240/2003	i, 10:12 AM		-				2
Albighutide — Powder and solvent for 005E-20 mFat 10 mR/h — Sobcutane	r solution for injection rous — 12/4/2023, 10:15 AM						
			11 million				
Aminosalicylic acid — Gauro-resista DOSA: 500 mg — Oral — Administer i	n granules. n case ut						
Paracetamol — Soluble tablet 0058: 1000 mg — Oral — 12/4/2023 :	ai 11:00 AM - 5:00 PM			_			
Forosemide — Oral suspension DOIX: 700 mg — Oral — 1244/2023 at	r 12:00 PM		_	_			
Brilliant green - Párit POSE 1000 mg - Curamenus - 12/4	2021. 10:19 AM						
			1000 mg				

Fig 194

Use the drop-down menu indicated in Fig 195 **A** to change the display mode.



The default display range is 8 hours. Click 12h or 24h to display the corresponding range. Fig 196 shows, as example, a detail of a 24 hours display.

		Tue 124	Decembe	•																			
37M	71, PM	12 AM	01 AM	82.AM	63 MM	:04 AM	(15 AM	26 AM	07 AM	08 AM	(19-20)	10.000	TT AM	12.244	01.HW	32.758	(13.7M)	04.88	85.PM	06.544	07.PM	88/M	:09 PM
Amin	salicylic	acid — 6	attris rasi	istant yrai	ini Aris																		
005	500 mg	- 0/4 -	Administ	10 P. 10 Case	ist.,																		
Barrie		- Soluble	and the second second																				
				a dayat	11:00-AM	5.00 PM	- anti-h	ethie not	ite :														
		Gral sus																					
uus.	700 mg	- Qrai -	1204/202	3 # 12:00	0.000																		

Fig 196 - 24 hours display (detail)

Use the left and right arrows (Fig 197 **A**) to scroll the screen contents back (left) and forward (right).



One click on the arrow button moves the screen for the time range specified in the field (i.e. if 8h is selected - as in the figure -, click the left arrow once to display the previous 8 hours, click the right arrow once to display the following 8 hours; if 12h is selected, the screen moves 12 hours per click).

Use the **NOW** button to display the current time again.

5.11. Treatment sorting buttons

The buttons indicated in Fig 198 **A** allow to sort and group the existing treatments by class, category or administration route.



The class, category and route of a treatment are defined during the configuration of the treatment.

COLOR AD VAN 2014							
Mon 4 December			- 00552				
-07 JUN	OB AM	09.888	10 AM	11 MM	12.PM	OT PW	102.Pt
Polithesantide — Eye drugs. DDD2: 500 mg — Dcular — 12/4/2023							
Altiglutide — Powder and solvent fo 0055-20 mFat 10 mFh — Solscitate							
			10 million				
Aminosalicylic acid — Santro-resista DOLL: 500 mg — Cral — Administre i	nt granules. n cave of						
Paracetamol — Sciuble tablet							
ODSE: 1000 mg - Oral - 12/4/2025.	at 11:00 AM - 5:00 PM						
6 15 M							
Forosemide — Oral surgersion ODSR: 700 mg — Oral — 12/4/2023 a	12:00 PM						
Brilliant green - Parit							
POSE 1000 mg - Eutanenus - 124	(2023, 10;19 AM		1001.00				
			the state				
IERAPY WEB					CLASS	CATEGORY ROUTE	15
GECUTION					1.L.H.D.	PROPERTY NUMBER	1.05

Click a button to sort the treatments according to the related sorting criterion. See, for example, in Fig 199, the treatments sorted by administration route (in the figure: Oral, Ocular, Subcutaneous, Cutaneous).

Click the button again to go back to the original sorting.

ETI I FENDINE																	NOW		240	
	Tue 12 D	redensor																		
PM TTPM	12.458	IN ANA IS	IZ AM	SAM SH	AM 15 4	M SEAM			TO A54	11.AM	12.74	STPM	DZ PM	22.054	04 PM	DE PM	OLPM.	177 1966	100 PM	13:PM
									0846											
Paracetamol -																				
005E 1000 mj	k - Dill -	2 times a	day at 115	0 AM - 5:0	09M → UN	I further no	0ca													
Forosemide																				
0051 100 mg	- Dral - 1	2/4/2023	at 12:00 PM																	
									CULAR											
									222411											
Polihexanide -			ton 10-13	ANT																
and the second	100000		CON UNIT																	
								3.80	TANÉOUT	ŧ.										
Albiglutide — i	Powder and	Lichwrit fi	ar salution	for opening	n															
0058-20 mi at	20 mlift -	Subrukan	eous - 12	1/6/2023.1	EB1 PM															
								Cut	ANTOUS											
Irillant green	Daniel																			
DOSE: 1000 mg			4/2023, 10																	
ERAPY WE	в														CLASS	CATE	3	ROUT	11.114	
ECUTION														10	ULASS	CALE		HOUT	E 65	

Fig 199

5.12. Compact view

Use the buttons indicated in Fig 200 **A** to switch to a more compact display mode (and back to normal). Fig 200 shows the normal view.

ten 4 December			- NEXT				
-07 300	OB AM	09.886	19 AM	TT MM	12.PM	ST PW	02.95
olihexanide — Eye drups 1002: 500 mg — Doulor — 12/4/2020	I, 10:12 AM						
Ubiglutide — Powder and solvent fo 1055-20 miliat 10 mil/h — Sobcutare	r solution for injection Ious — 12/4/2023, 10:15 AM	8					
			18 million				
<mark>uninosalicylic acid</mark> — Gantro-resista I <mark>CIE 500 mg</mark> — Oral — Administer I							
aracetamel — Sziuble tablet DOSE, 1000 mg — Oral — 12/4/2023	at 11:00 AM - 5:00 PM			_			
Geosemide — Oral suspension 2013: 700 mg — Oral — 12/4/2023 a	e 12:00 PM				_		
					-		
erBlast green – Paint 2058: 1000 mg – Cutanenus – 124	/2523.10:19.AM						
			1900 ang				

Fig 200

Fig 201 shows the compact view.

K NOW > Th	Tue 12 December						
	12.AM	04 AM	OK AM	14 PM	:D4 PM	III PM	1
Ammosalitylit acid							/~
DOSE 500 mg							_
Paracetamol DOLD: 1000 mg							~
Furgsemide DOLD: 700 mg							~
Politexanide DOSE 1900 mg						A	-
Abiglutide DOSE 20 ml at 20 ml/h						-	
Brillant groon DOSE: 1000 mg							~
							~
Norphise COSE: 10 mg in 300 ml at 10 mg/h							
Norphise COSE: 10 mg in 300 ml at 10 mg/h							
Morphise DOSE 10 mg is 300 ml at 10 mg/h							

Fig 201

Use the arrows on the right to enlarge the treatment rows and display the administration details and commands (see, for example, Fig 202).

< NOW > In	12.4	12 Decemb	er	D4.AM	IN AM			12.0	14		1141	PM.		-04	L PM	
Aminosalicylic acid DONE 500 mg		T.T.		50000	T. T.		1	T		1 0)			102	18
Paracetamol — Soluble tablet DOI:L 1060 mg — Oral — 2 times a day at 11:00 AM - 5:00 PM —	- until further	notica														
		1	sqP	10440800W 12/14/2023, 5:00 P	EXEC	VE										
				SUMMARY												
				мпылл нос 1000 mg 57												
											Pour	rihed by	süsev e	et Tion 2	Dec III, 101	ii.
Farosamide DOI:: 700 mg											Press	ritued by	ADBAIN O	n Tun Z	0er 85, 101	iii
											Press	nihed by .	ADMIN 0	et Tives 2	0er 85, 101	SFA
Politesanide											Press	rihed by	ADMIN O	et Tion, J	0ec 85, 101	ΈA
Polise 700 mg Polisesanide DOLE 1000 mg Abligituide											Press	nheg by .	süteri o	et Toos, J	0er 85, 101	'EA

Fig 202

5.13. Closed Therapy

If the Therapy is closed, a "lock" icon - a - is displayed on top of the Prescription module. See Fig 122 for an example.

On the Execution module, the only actions allowed on closed therapies are "Stop" and "Update" of running continuative administrations.